

### NAVIGATOR TELECOMMUNICATIONS, BEREIVED-DOCKETING DIV

8525 Riverwood Park Drive P.O. Box 13860 North Little Rock, AR 72113-0860 Phone: (501) 954-4000 Fax: (501) 954-4002

2007 APR 10 AM 11: 15

121

PUCO

April 9, 2007

07-401-TP-ATA

Public Utilities Commission of Ohio Attn: Docketing Division 180 East Broad Street Columbus, Ohio 43215-3793

RE: Local Tariff Revision Filing – Navigator Telecommunications, LLC. 90-9103-TP-TRF

Dear Docketing Division,

Enclosed for filing with the Public Utilities Commission of Ohio are revisions to Navigator Telecommunications, LLC.'s Tariff No. 1. Included are an original and 10 copies. Also enclosed are the required Telecommunication Application Form and all supporting documentation for this filing. This filing does not make any changes in rates. The purpose of this filing is to remove all minimum pricing from Tier 1 service and to eliminate all minimum/maximum pricing for Tier 2 services.

Please contact Mark Herring at (501) 954-4053 if you require more information about this filing. Thank you for your assistance in this matter.

Sincerely,

Michael McAlister General Counsel Navigator Telecommunications, LLC. (501) 954-4051 mike@navtel.com

This is to certify that the images appearing are an accurate and complete reproduction of a case file document delivered in the regular course of business Fechnician \_\_\_\_\_ Date Processed  $\frac{4-10-07}{2}$ 

# The Public Utilities Commission of Ohio **TELECOMMUNICATIONS APPLICATION FORM** (Effective: 10/01/2004) (Pursuant to Case Nos. 99-998-TP-COI and 99-563-TP-COI)

In the Matte	r of the Application of
Na	vigator Telecommunications, LLC. Case No. <u>D7</u> - <u>401</u> - <b>TP</b> - <u>ATA</u>
	prove Revisions to Local Exchange Tariff)
Mama af Da	gistrant(s) Navigator Telecommunications, LLC.
	Registrant(s) Same
	Registrant(s)
	/eb Addresswww.navtel.com
	Contact Person(s) Michael McAlister, General Counsel Phone (501) 954-4051 Fax (501)954-4002
	Contact Person's Email Address mike@navtel.com
<u> </u>	son for Annual Report Phone
	Contact Information Phone
Date <u>Ap</u>	ril 10, 2007 TRF Docket No CT-TRF or 90 - 9103 - TP-TRF
Mation for	$r_{\rm exc}$ and $r_{\rm exc}$ is already with filling $2 = V_{\rm exc} V_{\rm exc} V_{\rm exc}$
	protective order included with filing? $\Box$ Yes $\nearrow$ No
	waiver(s) filed affecting this case?  Yes XNo [Note: waiver(s) tolls any automatic timeframe] (ype (check all applicable): XCTS (IXC)  ILEC XCLEC  CMRS  AOS
Company	
	Other (explain)
<u>NOTE:</u> This	form must accompany all applications filed by telecommunication service providers subject to the Commission's rules promulgated
	998-TP-COI, as well as by ILECs filing an ARB or NAG case pursuant to the guidelines established in Case No. 96-463-TP-UNC. In
preferable <u>N</u>	OT to combine different types of filings, but if you do so, you must file under the process with the longest applicable review period.
L. Please	indicate the reason for submitting this form (check one)
	• —
• 2 (ABN)	Abandonment of all Services
	a. CLEC (90-day approval, 10 copies) 🛛 b. CTS (14-day approval, 10 copies) 🗖 c. ILEC (NOT automatic, 10 copies)
а 3 (ACE)	New Operating Authority for providers other than CMRS (30-day approval, 7 copies); for CMRS, see item No.15 on this page.
- 4 (4 C D)	$\Box$ a. Switched Local $\Box$ b. Non-switched local $\Box$ c. CTS $\Box$ d. Local and CTS $\Box$ e. Other (explain)
□ 4 (ACO) □ 5 (ACN)	
$\Box = 6$ (AEC)	Carrier-to-Carrier Contract Amendment to an agreement approved in a NAG or ARB case (30-day approval, 7 copies)
	NOTE: see item 25 (CTR) on page two of this form for all other contract filings.
□ 7 (AMT)	LEC Merger (30-day approval, 10 copies)
□ 8 (ARB)	Application for Arbitration (see 96-463-TP-COI for applicable process, 10 copies)
) (ATA)	Application for Tariff Amendment for Tier 1 Services, Application to Reclassify Service Among Tiers, or Change to Non-Tier Service
	<ul> <li>a. Tier 1 (and Carrier-to-Carrier tariff filings as set-forth in 95-845-TP-COI)</li> <li>i. Pre-filing submittal (30-day pre-filing submittal with Staff and OCC; Do Not Docket, 4 copies)</li> </ul>
	$\Box$ ii. New End User Service which has been preceded by a 30-day pre-filing submittal with Staff for all submittals and also wi
	OCC for Tier 1 residential services (0-day filing, 10 copies)
	iii. New End User Service (NOT preceded by a 30-day filing submittal, 30-day approval, 10 copies)
	□ iv. New Carrier-to-Carrier Service which has been preceded by a 30-day pre-filing with Staff (0-day filing, 10 copies)
	Xv. Change in Terms and Conditions, textual revision, correction of error, etc. (30-day approval, 10 copies)
	□ vi. Grandfather service (30-day approval, 10 copies) □ vii. Initial Carrier-to-Carrier Services Tariff subsequent to ACE approval (60-day approval, 10 copies)
	$\square$ viii. Withdrawal of Tier 1 service must be filed as an "ATW", not an "ATA" - see item 12, below
	b. Reclassification of Service Among Tiers (NOT automatic, 10 copies)
	C. Textual revision with no effect on rates for non-specific or non-tier service (30-day approval, 10 copies)
□ 10(ATC)	Application to Transfer Certificate (30-day approval, 7 copies)
ם 11 (ATR)	LEC Application to Conduct a Transaction Between Utilities (30-day approval, 10 copies)
🗆 12 (ATW)	
□ 13(CIO)	a. CLEC (60-day approval, 10 copies) <b>b.</b> ILEC (NOT automatic, 10 copies) Application for Change in Operations by Non-LEC Providers (0-day notice, 7 copies)
ם 14 (NAG)	Negotiated Interconnection Agreement Between Carriers (0-day effective, 90-day approval, 8 copies)
□ 15(RCC)	For CMRS providers only to Register or to Notify of a Change in Operations (0-day notice, 7 copies)
0 16 (SLF)	Self-complaint Application
	□ a. CLEC only -Tier 1 (60-day automatic, 10 copies)
- 10 (1) 100	□ b. Introduce or increase maximum price range for Non-Specific Service Charge (60-day approval, 10 copies)
□ 17(UNC)	Unclassified (explain) (NOT automatic, 15 copies) Tariff Application Involving only Tier 2 Services
a 18(ZTA)	NOTE: Notifications do not require or imply Commission Approval.
	□ a. New End User Service (0-day notice, 10 copies)
	□ b. Change in Terms and Conditions, textual revision, correction of error, etc. (0-day notice, 10 copies)
	c. Withdrawal of service (0-day notice, 10 copies)
🗆 19 Other	(explain) (NOT automatic, 15 copies)

#### THE FOLLOWING ARE TRF FILINGS ONLY, NOT NEW CASES (0-day notice, 3 copies)

- 20 Introduction or Extension of Promotional Offering
- □ 21 New Price List Rate for Existing Service □ a. Tier 1 □ b. Tier 2
- □ 22 Designation of Registrant's Process Agent(s)
- □ 23 Update to Registrant's Maps
- 24 Annual Tariff Option For Tier 2 Services indicate which option you intend to adopt to maintain the tariff. NOTE, changing options is only permitted once per calendar year.

□ Paper Tariff □ Electronic Tariff. If electronic, provide the tariff's web address:

THE FOLLOWING ARE CTR FILINGS ONLY, NOT NEW CASES (0-day notice, 7 copies)

□ 25 Application to establish, revise, or cancel an end-user contract. (*NOTE: see item 6 on page 1 of this form for carrier-to-carrier contract amendments*) CTR Docket No.\_\_\_\_\_\_. - TP - CTR (Use same CTR number throughout calendar year)

II. Please indicate which of the following exhibits have been filed. The numbers (corresponding to the list on page (1) and above) indicate, at a minimum, the types of cases in which the exhibit is required:

Ö	[all]	A copy of any motion for waiver of O.A.C. rule(s) associated with this filing. NOTE: the filing of a motion for waiver tolls
		any automatic timeframe associated with this filing.
0	[3]	Completed Service Requirements Form.
	[3, 9(vii)]	A copy of registrant's proposed tariffs. (Carrier-to-Carrier resale tariff also required if facilities-based)
D	[3]	Evidence that the registrant has notified the Ohio Department of Taxation of its intent to conduct operations as a telephone utility in the State of Ohio.
D	[3]	Brief description of service(s) proposed.
0	[3a-b,3d]	Explanation of whether applicant intends to provide $\Box$ resold services, $\Box$ facilities-based services, or $\Box$ both resold and facilities-
	[54-0,54]	based services.
0	[3a-b,3d]	Explanation as to whether CLEC currently offers CTS services under separate CTS authority, and whether it will be including
-	[****,***]	those services within its CLEC filing, or maintaining such CTS services under a separate affiliate.
0	[3a-b,3d]	Explanation of how the proposed services in the proposed market area are in the public interest.
0	[3a-b,3d]	Description of the proposed market area.
	[3a-b,3d]	Description of the class of customers (e.g., residence, business) that the applicant intends to serve.
0	[3a-b,3d]	Documentation attesting to the applicant's financial viability, including the following:
	104 0,04	1) An executive Summary describing the applicant's current financial condition, liquidity, and capital resources.
		Describe internally generated sources of cash and external funds available to support the applicant's operations that
		are the subject of this certification application.
		2) Copy of financial statements (actual and pro forma income statement and a balance sheet). Indicate if financial
		statements are based on a certain geographical area(s) or information in other jurisdictions
		3) Documentation to support the applicant's cash an funding sources.
	[3a-d]	Documentation attesting to the applicant's technical and managerial expertise relative to the proposed service offering(s) and
-	()	proposed service area.
	[3a-d]	Documentation indicating the applicant's corporate structure and ownership.
	[3a-b,3d]	Information regarding any similar operations in other states. Also, if this company has been previously certified in the State of
-		Ohio, include that certification number.
	[3a-b,3d]	Verification that the applicant will maintain local telephony records separate and apart from any other accounting records in
		accordance with the GAAP.
	[3a-b,3d]	Verification of compliance with any affiliate transaction requirements.
	[3a-b,3d]	Explanation as to whether rates are derived through (check all applicable):
		□ interconnection agreement, □ retail tariffs, or □ resale tariffs.
D	[1,3a-b,3d]	Explanation as to which service areas company currently has an approved interconnection or resale agreement.
	[3a-b,3d, 9a(i-iii)]	Explanation of whether applicant intends to provide Local Services which require payment in advance of
		Customer receiving dial tone.
D	[3a,3b,3d,	Tariff sheet(s) listing the services and associated charges that must be paid prior to customer receiving dial tone (if applicable).
	9a,(i-iii)]	
D	[3a-b,3d,8]	Letters requesting negotiation pursuant to Sections 251 and 252 of the Telecommunications Act of 1996 and a proposed timeline for construction, interconnection, and offering of services to end users.
D	[3-5,7,10-11,13]	Certification from Ohio Secretary of State as to party's proper standing (domestic or foreign corporation, authorized use of
u	[5-5,7,10-11,15]	fictitious name, etc.). In transfer of certificate cases, the transferee's good standing must be established.
٥	[3-4,7,10-11,13]	List of names, addresses, and phone numbers of officers and directors, or partners.
	[3]	A sample copy of the customer bill and disconnection notice the applicant plans to utilize.
	[1,4,9,10-13,16-21]	Copy of superseded tariff sheet(s) & price list(s), if applicable, marked as Exhibit A.
X	[1,4,9,10-13,16-21]	Copy of revised tariff sheets & price lists, marked as Exhibit B.
	[3]	Provide a copy of any customer application form required in order to establish residential service, if applicable.
X	[1-2,4-7,9,12-	Description of and rationale for proposed tariff changes, including a complete description of the service(s) proposed or affected.
1~	13,16,18-23,25]	Specify for each service affected whether it is $\Box$ business; $\Box$ residence; or $\Delta$ both. Also indicate whether it is a $\Delta$ switched or $\Box$
	10,10,10 <sup>-</sup> 40,40 <sup>-</sup>	dedicated service. Include this information in either the cover letter or Exhibit C.

	21]	□ Tier 1 price list increases <b>must</b> be within an approved range of rates.
		SLF Filings – Do NOT send customer notice until it has been reviewed and approved by Commission Staff
X	[2,4-5,9a(v),	Copy of real time notice which has been/will be provided to customers.
	9b, 10,12-13,16,	NOTE: SLF Filings - Do NOT send customer notice until it has been reviewed and approved by Commission Staff
	18(b-c),20-21]	No notice requirement
X	[1,2,5,9a(v),11-13,	Affidavit attesting that customer notice has been provided.
	18, 21(increase	Not was in it
<u> </u>	only)]	Not required Copy of Notice which has been provided to ILEC(s).
	[2,12]	Copy of Notice which has been provided to ILEC(s).
	[2,12]	Listing of Assigned (NPA) NXX's where in the LECs (NPA) NXX's would be reassigned.
	[2,4,10,12-13,]	List of Ohio exchanges specifically involved or affected.
	[14]	The interconnection agreement adopted by negotiation or mediation.
	[15]	For commercial mobile radio service providers, a statement affirming that registrant has obtained all necessary federal authority
		to conduct operations being proposed, and that copies have been furnished by cellular, paging, and mobile companies to this
		Commission of any Form 401, 463, and / or 489 which the applicant has filed with the Federal Communications Commission.
	[15]	Exhibits must include company name, address, contact person, service description, and evidence of registration with the Ohio
		Secretary of State.
	[24]	Affidavit that total price of contract exceeds total cost of all regulated services.
	[5,13]	New title sheet with proposed new company name.
	[1,3,13]	For CLECs, List of Ohio Exchanges the applicant intends to serve (Use spreadsheet from:
		http://www.puc.state.oh.us/puco/forms/form.cfm?doc_id=357).
	[1,3a-b,3d,7,	Maps depicting the proposed serving and calling areas of the applicant.
	10,13, 23]	If Mirroring Large ILEC exchanges for both serving area and local calling areas: • Serving area must be clearly reflected
	1	on an Ohio map attached to tariffs and textually described in tariffs by noting that it is reflecting a particular large
		ILEC/CLEC territory, and listing the involved exchanges. • Local calling areas must be clearly reflected on an Ohio map
		attached to the tariffs, and/or clearly delineated in tariffs, including a complete listing of each exchange being served and all
		exchanges to which local calls can be made from each of those exchanges.
}		If Self-defining serving area and/or local calling area as an area other than that of the established ILEC exchange(s): •
		Serving Area must be clearly reflected on an Ohio map attached to the tariffs, and textually described in tariffs by listing the
		involved exchanges. • Local Calling Areas must be described in the tariff through textual defineation and clear maps. Maps
		for self-defined serving and local calling areas are required to be traced on United States Geological Survey topography
		maps. These maps are the Standard Topographic Quadrangle maps, 7.5 minute 1:24,000.
	<u> </u>	Other information requested by the Commission staff.
	[3]	Initial certification that includes Tier 2 Services, indicate which option you intend to adopt to maintain the tariff:
		Paper Tariff  Electronic Tariff - If electronic, provide the web address for the tariff:
	· · · · · · · · · · · · · · · · · · ·	

III. Registrant hereby attests to its compliance with the following requirements in the Service Requirements Form, as well as all pertinent entries and orders issued by the Commission with respect to these issues. Further, registrant hereby affirms that it will maintain with its TRF docket an up-to-date, properly marked, copy of the Service Requirements Form available for public inspection.

#### MANDATORY REQUIREMENTS FOR ALL BASIC LOCAL EXCHANGE AND CTS PROVIDERS:

[x] Sales tax

[x] Minimum Telephone Service Standards (MTSS)

[x] Surcharges

#### MANDATORY REQUIREMENTS FOR ALL BASIC LOCAL EXCHANGE PROVIDERS:

[x] 1+ IntraLATA Presubscription

#### SERVICE REQUIREMENTS FOR PROVISION OF CERTAIN SERVICES (CHECK ALL APPLICABLE):

Discounts for Persons with Communication Disabilities and the Telecommunication Relay Service [Required if toll service provided]

- Emergency Services Calling Plan [Required if toll service provided]
- □ Alternative Operator Service (AOS) requirements [Required for all providing AOS (including inmate services) service]
- Limitation of Liability Language [Required for all who have tariff language that may limit their liability]
- D Termination Liability Language [Required for all who have early termination liability language in their tariffs]
- □ Service Connection Assistance (SCA) [Required for all LECs]

□ Local Number Portability and Number Pooling [Required for facilities-based LECs]

D Package Language [Required for tariffs containing packages or service bundles containing both local and toll and/or non-regulated services]

- IV. List names, titles, phone numbers, and addresses of those persons authorized to respond to inquiries from the Consumer Services Department on behalf of the applicant regarding end-user complaints:
- V. List names, titles, phone numbers, and addresses of those persons authorized to make and/or affirm or verify filings at the Commission on behalf of the applicant:

<u>NOTE</u>: An annual report is required to be filed with the Commission by each company on an annual basis. The annual report form will be sent for completion to the address and individual(s) identified in this Section unless another address or individual is so indicated.

VI. List Name(s), DBA(s) and PUCO Certification Number(s) of any affiliates you have operating in Ohio under PUCO authority, whether Telecommunication or other. (If needed, use a separate sheet and check here:

### **AFFIDAVIT**

#### **Compliance with Commission Rules and Service Standards**

I am an officer of the applicant corporation, <u>Navigator Telecommunications, LLC.</u>, and am authorized to make this statement (Name of Company) on its behalf. I attest that these tariffs comply with all applicable rules, including the Minimum Telephone Service Standards (MTSS) for the state of Ohio. I understand that tariff notification filings do not imply Commission approval and that the Commission's rules, including the Minimum Telephone Service Standards, as modified and clarified from time to time, supersede any contradictory provisions in our tariff. We will fully comply with the rules of the state of Ohio and understand that noncompliance can result in various penalties, including the suspension of our certificate to operate within the state of Ohio.

I declare under penalty of perjury that the foregoing is true and correct.

04/09/07 at Little Rock, ARKANS (Date) (Location) Executed on & - (TENERAL ) anDEZ (Date)

\* This affidavit is required for every tariff-affecting filing. It may be signed by counsel or an officer of the applicant, or an authorized agent of the applicant.

MILITAEL MCALISTER

NYY LLYNN YNY YN YWLYYDD YLYNYN HID HLYNNYDD LLYNN YD LLYNN MAD DDDNYD HD HEANTARD DDD

#### **VERIFICATION**

I, <u>PlicitAELINCHUSTER</u> verify that I have utilized, verbatim, the Commission's Telecommunications Application Form and that all of the information submitted here, and all additional information submitted in connection with this case, is true and correct to the best of my knowledge.

Conser 04/09/07 (Date)

\*Verification is required for every filing. It may be signed by counsel or an officer of the applicant, or an authorized agent of the applicant.

Send your completed Application Form, including all required attachments as well as the required number of copies, to:

Public Utilities Commission of Ohio Attention: Docketing Division (or to the Telecommunications Division Chief if a prefiling submittal) 180 East Broad Street, Columbus, OH 43215-3793 Exhibit A

.-

## CHECK SHEET

The Title Page and pages listed below of this tariff are effective as of the date shown. Revised sheets contain all changes from the original tariff that are in effect as of the date indicated.

PAGE	REVISION	PAGE	<u>REVISION</u>
Title Sheet	Original	31	Original
1	Fourth Revised *	32	Original
1.1	First Revised *	33	Original
1.2	Third Revised *	34	Original
1.3	Third Revised *	35	Original
2	Original	36	Original
3	Original	37	Original
4	Second Revised *	38	Original
5	Original	39	Original
6	Second Revised *	40	Original
7	Original	41	Original
8	Original	42	Original
9	Original	43	Original
10	Original	44	Original
11	Original	45	Original
12	Original	46	Original
13	Original	47	Original
14	Original	48	Original
15	Original	49	Original
16	Original	50	Original
17	Original	51	Original
18	Original	52	Original
19	Original	53	Original
20	Original	54	Original
21	Original	55	Original
22	Original	56	Original
23	Original	57	Original
24	Original	58	Original
25	Original	59	Original
26	Original	60	Original
27	Original	61	Original
28	Original	62	Original
29	Original	63	Original
30	Original	64	Original

8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 ...

## CHECK SHEET

The Title Page and pages listed below of this tariff are effective as of the date shown. Revised sheets contain all changes from the original tariff that are in effect as of the date indicated.

PAGE	<u>REVISION</u>	PAGE	<b>REVISION</b>
65	Original	99	Original
66	Original	100	Original
67	Original	101	Original
68	Original	102	Original
69	Original	103	Original
70	Original	104	Original
71	Original	105	Original
72	Original	106	Original
73	Original	107	Original
74	Original	108	Original
75	Original	109	Original
76	Original	110	Original
77	Original	111	Original
78	Original	112	Original
79	Original	113	Original
80	Original	114	Original
81	Original	115	Original
82	Original	116	First Revised *
83	Original	117	Original
84	Original	118	Original
85	Original	119	Original
86	Original	120	Original
87	Original	121	Original
88	Original	122	First Revised *
89	Original	123	Original
90	Original	124	Original
91	Original	125	Original
92	Original	126	Original
93	Original	127	Original
94	Original	128	Original
95	Original	129	Original
96	Original	130	Original
97	Original	131	Original
98	Original	132	Original

Issued: October 5, 2006 Effective: October 6, 2006 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC.

8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

### CHECK SHEET

The Title Page and pages listed below of this tariff are effective as of the date shown. Revised sheets contain all changes from the original tariff that are in effect as of the date indicated.

PAGE	<u>REVISION</u>	PAGE	<u>REVISION</u>
133	First Revised *	158	Original
134	Original	159	Original
135	Original	160	Original
136	First Revised *	161	Original
137	First Revised *	162	Original
138	Original	163	Original
139	Original	164	Original
140	Original	165	Original
141	Original	166	Original
142	Original	167	Original
142.1	Original	168	Original
142.2	Original	169	Original
142.3	Original	170	Original
142.4	Original	171	Original
142.5	Original *	172	Original
142.6	Original *	173	Original
142.7	Original *	174	Original
142.8	Original *	175	Original
142.9	Original *	176	Original
143	Original	177	Original
144	Original	178	Original
145	Original	179	Original
146	Original	180	Original
147	Original	181	Original
148	Original	182	Original
149	Original	183	Original
150	Original	184	Original
151	Original	185	Original
152	Original	186	Original
153	Original	187	Original
154	Original	188	Original
155	First Revised	189	Original
156	First Revised	190	Original
157	First Revised	191	Original

 Issued: October 5, 2006
 Effective: October 6, 2006

 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002
 June 20, 2002

 Issued by:
 Louis F. McAlister, Jr., President

 Navigator Telecommunications, LLC.
 Navigator Telecommunications, LLC.

8525 Riverwood Park Drive, P.O. Box 13860

North Little Rock, AR 72113-0860

...

## CHECK SHEET

The Title Page and pages listed below of this tariff are effective as of the date shown. Revised sheets contain all changes from the original tariff that are in effect as of the date indicated.

PAGE	<b>REVISION</b>	PAGE	REVISION
192	Original		
193	Original		
194	Original		
195	Original		
195.1	Original *		
196	Original		
197	Original		
198	Original		
199	Original		
200	Original		
201	First Revised		
202	First Revised		
203	Original		
204	Original		
205	Original		
206	First Revised *		
207	Original		
208	Original		
209	First Revised *		
210	First Revised *		
211	Original		
212	Original		
213	Original		
214	First Revised		
215	First Revised		
216	First Revised		
217	Original		
218	Original		
219	Original		
220	Original		
221	Original		
222	Original		
223	Original *		

 Issued:
 October 5, 2006
 Effective:
 October 6, 2006

 In accordance with Case No. 00-21-TP-ACE, issued
 June 20, 2002
 .

 Issued by:
 Louis F. McAlister, Jr., President

 Navigator Telecommunications, LLC.
 8525 Riverwood Park Drive, P.O. Box 13860

 North Little Rock, AR 72113-0860

## **3.2** Service Charges, cont. (1)

#### 3.2.2 Nonrecurring Charges

A. Application of Nonrecurring Charges to Establish Services

Except where otherwise specified in this paragraph, the same nonrecurring charges apply to residence and non-residence service.

The Central Office Connection and Line Connection Charges do not apply to the transfer of an existing central office line servicing a reseller's end user when there is no interruption of service.

1. Exchange Service

The following charges are applicable to each individual exchange service being established:

e	Complex		Sin	nple
_	Minimum	Maximum	Minimum	Maximum
a. Local Exchange Service	;			
- Service Ordering Char	ge			
per location, per occasi	ion			
- Residence	\$12.50	\$25.50	\$12.50	\$25.50
- Non Residence	10.85	22.50	18.50	35.50
- Central Office Connec Charge, per terminatio				
- Residence	7.00	12.50	7.00	12.50
- Non Residence	14.00	25.00	10.00	18.50
- Line Connection Charge per termination	ge			
- Residence	7.00	12.50	7.00	12.50
- Non Residence	12.00	22.50	20.00	32.50

Issued: June 25, 2002			Effective: June 25, 2002
In accordance with Case N	No. 00-21-TP-ACE, issued	June 20, 2002	<u> </u>
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommun	ications, LLC.	
	8525 Riverwood Park Drive,	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

# 3.2 Service Charges, cont. (1)

## 3.2.2 Nonrecurring Charges

- A. Application of Nonrecurring Charges to Establish Service, cont.
  - 1. Exchange Service, cont.

The following charges are applicable to each individual exchange service being established:

-	Complex		Simple	
-	Minimum	Maximum	<u>Minimum</u>	Maximum
b Central Office Connec	tion			
Charge, per termination	n			
- Residence	-	-	-	-
- Non Residence	-	-	11.00	18.00
- Line Connection Char	ge,			
per termination (includ				
installation of telephon	ie)			
- Residence	-	-	-	-
- Non Residence	-	-	52.00	74.50
c. Centrex stations and ter	minals			
Exhibition Hall stations	s, Airport			
Switching system static	ns			
- Service Ordering Char	ge			
per location, per occasi	ion			
- Residence	-	-	-	-
- Non Residence	38.00	54.80	-	-
- Central Office Connec	tion			
Charge, per station terr	nination			
- Residence	-	-	-	-
- Non Residence	1.50	4.50	-	-

Issued: June 25, 2002			Effective: June 25, 2002
In accordance with Case No	0. 00-21-TP-ACE, issued	June 20, 2002	
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommun	ications, LLC.	
8.	525 Riverwood Park Drive	P.O. Box 13860	د
	North Little Rock, AR	72113-0860	

#### 3.2 Service Charges, cont. (1)

### 3.2.2 Nonrecurring Charges, cont.

- A. Application of Nonrecurring Charges to Establish Service, cont.
  - 1. Exchange Service, cont.

Charges applicable to each individual exchange service being established:

	Complex		Simple	
_	<u>Minimum</u>	Maximum	<u>Minimum</u>	Maximum
<ul> <li>Line Connection Charge per station termination</li> <li>Residence</li> <li>Non Residence</li> </ul>	- 1.50	5.50	-	-
<ul> <li>d. PBX Trunks</li> <li>- Service Ordering Charper location, per occasi</li> </ul>	~			
- Residence - Non Residence	18.50 18.50	32.50 32.50	-	- -
<ul> <li>Central Office Connec Charge, per termination</li> <li>Residence</li> <li>Non Residence</li> </ul>		20.50 20.50	-	-
- Line Connection Charg per termination	ge, 8.35	13.35	-	-

(D)

(1) Current prices for these services are shown in Section 12.2 of this tariff.

Issued: October 5, 2006 Effective: October 6, 2006 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 . Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

••

# SECTION 3 - RATES – LOCAL EXCHANGE SERVICES, CONT.

# 3.2 Service Charges, cont. (1)

# 3.2.2 Nonrecurring Charges

# B. Application of Nonrecurring Charges for Restoral of Service

	Com	plex	Simple		
-	Minimum	Maximum	<u>Minimum</u>	Maximum	
<ol> <li>Service restored after temporary denial but prior to completion of order to discontinue service, per service or system Restoral Charge         <ul> <li>Residence</li> <li>Non Residence</li> </ul> </li> </ol>	\$22.50 22.50	\$35.50 35.50	\$26.30 26.30	\$38.30 38.30	
	22.30	55.50	20.50	56.50	
2. Service restored after temporary suspension Restoral Charge					
- Residence	55.00	69.50	28.00	39.50	
- Non Residence	65.00	83.50	28.00	39.50	
3. Service restored after temporary interception Restoral Charge					
- Residence	55.00	69.50	28.00	39.50	
- Non Residence	75.00	98.50	28.00	39.50	

Issued: June 25, 2002			Effective: June 25, 2002
In accordance with Cas	se No. 00-21-TP-ACE, issued	June 20, 2002	
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommun	ications, LLC.	
	8525 Riverwood Park Drive	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

# 3.3 Local Service Rates (1)

#### 3.3.1 Flat Rate Service

Flat Rate service consists of a usage package that includes an unlimited number of local messages.

**..** 

A.	Rates and Charges	Usage Package	Monthly Rate
		<u>Minimum</u>	<u>Maximum</u>
	Residence PBX Trunks	\$6.50	\$11.50
	Residence Individual	\$5.00	\$10.00

## 3.3.2 Local Service Rates

A.	Schedule of Monthly Rates – Network Access Lines
----	--

Description	Access Area	A	twork ccess Line		tral fice ination		tal nthly arge
-		Min	Max	Min	Max	Min	Max
Non - Residence							
- Non-Rotary	Α	-	-	-	-	-	-
With Touch-Tone	В	13.95	17.95	.30	4.30	14.25	22.25
	С	15.95	19.95	.30	4.30	16.25	24.25
	D	18.45	22.45	.30	4.30	18.75	26.75
- Rotary	А	-	_	-	-	-	_
- PBX Trunk	В	13.95	17.95	4.00	8.00	17.95	25.95
With Touch-Tone	С	15.95	19.95	4.00	8.00	19.95	27.95
	D	18.45	22.45	4.00	8.00	22.45	30.45

Issued: June 25, 2002			Effective: June 25, 2002
In accordance with Case No. 00	-21-TP-ACE, issued	June 20, 2002	<b>^</b>
Issued by:	Louis F. McAlister, Jr.,	President	
Ν	lavigator Telecommuni	cations, LLC.	
8525	Riverwood Park Drive,	P.O. Box 13860	
]	North Little Rock, AR ´	72113-0860	

# 3.3 Local Service Rates, cont. (1)

# 3.3.2 Local Service Rates, cont.

# A. Schedule of Monthly Rates – Network Access Lines, cont.

D. I. I	Access	Ac	twork cess		fice		nthly
Description	Area		.ine		ination		arge
Non - Residence		Min	Max	Min	Max	Min	Max
- Non-Rotary	A	-	-	-	-	-	-
Without	В	13.95	17.95	.30	4.30	14.25	22.25
Touch-Tone	С	15.70	19.70	.30	4.30	16.00	24.00
	D	15.70	19.70	.30	4.30	16.00	24.00
- Rotary	A	-	-	-	_	-	-
- PBX Trunk	В	13.95	17.95	4.00	8.00	17.95	25.95
Without	С	15.70	19.70	4.00	8.00	19.70	27.95
Touch-Tone	D	15.70	19.70	4.00	8.00	19.70	27.95
Residence							
- Individual	А	-	-	-	-	-	-
With Touch-Tone	B	2.40	6.40	.30	4.30	2.70	10.70
	$\mathbf{C}$	2.40	6.40	.30	4.30	2.70	10.70
	D	2.40	6.40	.30	4.30	2.70	10.70
- PBX Trunk	А	-	_	-	-	-	-
With Touch-Tone	B	13.95	17.95	4.00	8.00	17.95	25.95
	С	15.95	19.95	4.00	8.00	19.95	27.95
	D	18.45	22.45	4.00	8.00	22.45	30.45

Issued: June 25, 200	2		Effective: June 25, 2002
In accordance with C	ase No. 00-21-TP-ACE, issued	June 20, 2002	
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommuni	cations, LLC.	
	8525 Riverwood Park Drive,	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

### 3.3 Local Service Rates, cont. (1)

# 3.3.2 Local Service Rates, cont.

A. Schedule of Monthly Rates – Network Access Lines, cont.

		Net	twork	Cen	tral	To	tal
	Access	A	cess	Of	fice	Mor	nthly
Description	Area	I	ine	Term	ination	Cha	arge
		Min	Max	Min	Max	Min	Max
Residence							
- Individual	А	-	-	-	**	-	-
Without	В	2.40	6.40	.30	4.30	2.70	10.70
Touch-Tone	С	2.40	6.40	.30	4.30	2.70	10.70
	D	2.40	6.40	.30	4.30	2.70	10.70
- PBX Trunk	A	· _	-	-	-	-	-
Without	В	13.95	17.95	4.00	8.00	17.95	25.95
Touch-Tone	С	15.70	19.70	4.00	8.00	19.70	27.70
	D	15.70	19.70	4.00	8.00	19.70	27.70

### 3.3.3 Measured Rate Services

A. Local Message Charge Schedule

	Initial	Minute	Additional Minute		
Rate Mileage	or Fractio	n Thereof	or Fraction Thereof		
	Min.	Max.	Min.	<u>Max</u>	
0 - 10	\$.0253	\$.0453	\$.007	\$.01	
11 - 22	.0297	.0497	.01	.02	
23 and over	.0342	.0542	.0125	.02	

Issued:	June 25, 2002	Effective: June 25, 2002
In acco	dance with Case No. 00-21-TP-ACE, issuedJune 20, 2002	·
Issued l	by: Louis F. McAlister, Jr., President	
	Navigator Telecommunications, LLC.	
	8525 Riverwood Park Drive, P.O. Box 13860	
	North Little Rock, AR 72113-0860	

### 3.3 Local Service Rates, cont. (1)

#### 3.3.4 Local Calling Plus

Measured Rate Service shown above is not required to take advantage of Local Calling Plus. It is available to all measured, message and flat rate service residence and non-residence customers. Local usage rates are shown below.

#### A. Local Calling Plus Charge Schedule

	Initial	Minute	Additional Minute		
Rate Mileage	or Fractic	n Thereof	or Fraction Thereof		
	Min.	Max.	Min.	Max	
0 - 10	\$.0290	\$.0490	\$.005	\$.02	
11 - 22	.0330	.0530	.01	.025	
23 and over	.0390	.0590	.01	.03	

\$

# SECTION 3 - RATES – LOCAL EXCHANGE SERVICES, CONT.

# 3.4 Operator Service Rates – Traditional (1)

3.4.1 Service Charges

		Minimum Per Use <u>Charge</u>	Maximum Per Use <u>Charge</u>
	Automated Calling Card	\$.30	\$1.00
	Station-to-Station		
	Customer Dialed-Operator	.80	3.00
	Assisted-Calling Card		
	Station-to-Station		
	Operator Assisted	.75	2.75
	Station-to-Station		
	Operator Assisted	1.50	6.00
	Person-to-Person	75	2.00
	Operator Assisted	.75	3.00
	Third Number Billed		
3.4.2	Directory Assistance Service - Local		
	Residence Charge		
	Where customer direct dials	\$.15	\$.60
	directory assistance number		
	Where customer places call to directory	.15	.90
	assistance attendant via an operator	.15	.90
	assistance attendant via an operator		
	Non - Residence Charge		
	Where customer direct dials	.15	.60
	directory assistance number		
	Where customer places call to directory	.15	.90
	assistance attendant via an operator	.15	.70
	assistance attendant via an operator		
	National Directory Assistance		
	Direct Dialed	. <b>6</b> 0	1.50
	Alternate Billed Calls	.60	1.50

Issued: June 25, 2002			Effective: June 25, 2002
In accordance with Cas	se No. 00-21-TP-ACE, issued	June 20, 2002	'
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommun	ications, LLC.	
	8525 Riverwood Park Drive	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

## 3.4 Operator Service Rates – Traditional, cont. (1)

# 3.4.3 Information Call Completion (ICC) Service

...

A. The following rates and charges are for automatic or operator call completion of a directory assistance call after receipt of the requested number. If a customer selects an alternate billing option, i.e., calling card, billed-to-third party, collect and person-to-person, special handling, service charges as set forth above apply in addition to the ICC charge.

	Minimum	Maximum
	Per Use	Per Use
	Charge	Charge
ICC, per call	.15	.75

## 3.4.4 Busy Line Verify and Line Interrupt Service

-	Minimum	Maximum
	Per Use	Per Use
	<u>Charge</u>	<u>Charge</u>
Verification	.60	3.00
Interrupt	.80	4.00

Issued: June 25, 2002			Effective: June 25, 2002
In accordance with Cas	e No. 00-21-TP-ACE, issued	June 20, 2002	<u> </u>
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommun	ications, LLC.	
	8525 Riverwood Park Drive,	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

# 3.5 Alternative Operator Service Rates (1)

# 3.5.1 Prices

..

## 1. Operator Assisted Services

	Minimum Nonrecurring	Maximum Nonrecurring
Description	Charge	Charge
Automated Calling Card Station-to-Sta Per call	tion, \$0.30	\$ 0.70
Customer-Dialed Calling Card Station-to-Station, per call	.75	1.70
Operator-Assisted Station-to-Station, Per call	.90	2.50
Operator-Assisted Person-to-Person, Per call	1.00	4.80
Operator-Assisted Third Number Bille Per call	d, 1.00	2.50

Issued: June 25, 2002			Effective: June 25, 2002
In accordance with Case	No. 00-21-TP-ACE, issued	June 20, 2002	
Issued by:	Louis F. McAlister, Jr.,	, President	
	Navigator Telecommuni	cations, LLC.	
	8525 Riverwood Park Drive,	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

## 3.7 Call Blocking (1)

..

# 3.7.1 Rates and Charges

1. Residence and Non-Residence Customer Requested Call Blocking

The following rates and charges will be waived for the initial establishment of call blocking. These charges will apply to subsequent requests for blocking.

Minimum Nonrecurring	Maximum Nonrecurring	-
Charge	Charge	<u>USOC</u>
\$ 5.30	\$15.00	RTVXN
	15.00	RTVXN
	65.00	
tions 30.30	65.00	RTVXN
king		
es apply to spons	or requested call b	locking:
, 45.00	90.00	NR9CB
nor		
per		
45.00	90.00	NR9CB
, per		
45.00	90.00	NR9CB
	Nonrecurring <u>Charge</u> \$ 5.30 ual 5,30 tions 30.30 king ges apply to spons , 45.00 per 45.00 , per	Nonrecurring ChargeNonrecurring Charge\$ 5.30\$15.00ual e $5.30$ $15.00$ ual e $5.30$ $65.00$ utions 30.30 $65.00$ king $65.00$ king $90.00$ per $45.00$ $90.00$ per $45.00$ $90.00$

Issued: June 25, 2002			Effective: June 25, 2002
In accordance with Case No.	00-21-TP-ACE, issued	June 20, 2002	
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommun	ications, LLC.	
852	25 Riverwood Park Drive	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

..

# SECTION 3 - RATES - LOCAL EXCHANGE SERVICES, CONT.

#### 3.8 Toll Restriction (1)

#### 3.8.1. Rates and Charges

The following rates and charges apply to toll restriction service and are in addition to all other rates and charges applicable to residence, non-residence, or PBX trunk service. See Section 3.3 for local service rates for these services.

••	Minimum Nonrecurring Charge	Maximum Nonrecurring Charge	USOC
Toll Restriction, per individual residence service	\$10.00	\$30.00	RTVX5
Toll Restriction, per individual non-residence service or PBX trunk equipped	\$20.00	\$44.00	KXT
	Minimum Monthly <u>Charge</u>	Maximum Monthly <u>Charge</u>	USOC
Toll Restriction, per individual residence service	\$2.00	\$12.00	RTVX5
Toll Restriction, per individual non-residence service or PBX trunk equipped	\$44.00	\$80.00	KXT

(1) Current prices for these services are shown in Section 12.8 of this tariff.

Note: A toll restriction charge will not apply to those residential customers: 1) who elect toll restriction in lieu of a deposit, 2) who elect toll restriction while making payments for a final bill, or 3) who elect payment arrangements and toll restriction in lieu of denial.

Issued: June 25, 2002		· · · ·	Effective: June 25, 2002
In accordance with Case No	. 00-21-TP-ACE, issued _	June 20, 2002	<b>*</b>
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommun	ications, LLC.	
8	525 Riverwood Park Drive	, P.O. Box 13860	
	North Little Rock, AR	72113-0860	

## 3.9 Custom Calling Features (1)

#### 3.9.1 Residence - Rates and Charges

	Minimum	Maximum
	Per Month	Per Month
Description /Billing Code/	Charge	Charge
Call Waiting /ESX/	\$2.50	\$9.40

#### 3.9.2 Business - Rates and Charges

	Minimum	Maximum
	Per Month	Per Month
Description /Billing Code/	Charge	Charge
Call Waiting /ESX/	\$2.50	\$18.20

(1) Current prices for these services are shown in Section 12.9 of this tariff.

Issued: October 5, 2006 Effective: October 6, 2006 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 . Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 (D) |

(D)

(D)

(D)

## 3.10 Advanced Custom Calling Features (1)

# 3.10.1 Residence - Rates and Charges

	Description /Billing Code/	Minimum Per Month Charge	Maximum Per Month <u>Charge</u>	
	Repeat Dialing /NSS/	\$2.50	\$5.00	
	Caller ID /NSD/	2.50	9.00	
	Caller ID w/Name /NMP/ (See Note)	1.25	5.00	
	Automatic Callback /NSQ/	2.50	5.00	
	Call Screening /NSY/	2.50	8.00	
		Minimum	Maximum	
		Per Use	Per Use	
	Description	Charge	Charge	
	Repeat Dialing	\$.50	\$1.50	
	Automatic Callback	.50	1.50	
	Call Trace, per successful activation	2.50	4.75	
3.10.2	Non-Residence - Rates and Charges			
		Minimum Per Month	Maximum Per Month	
	Description /Billing Code/	<u>Charge</u>	Charge	
	Caller ID /NSD/	2.50	9.00	1) 1) 1)
	Call Waiting ID /NWT/	1.00 Minimum Per Use	4.00 Maximum Per Use	(1
	Description	Charge	Charge	
		<i>Q</i>	<b></b>	()
	Call Trace, per successful activation	2.50	4.75	

Note: The monthly rate for Caller ID with Name is in addition to the monthly rate for Caller ID.

Issued: October 5, 2006		Effective: October 6, 2006
In accordance with Case N	lo. 00-21-TP-ACE, issued	June 20, 2002
Issued by:	Louis F. McAlister, Jr.	, President
	Navigator Telecommuni	cations, LLC.
	8525 Riverwood Park Drive,	P.O. Box 13860
	North Little Rock, AR	72113-0860

#### 3.11 Central Office Services (1)

..

3.11.1 Direct Inward Dialed (DID) Service

#### A. Regulations

DID service will be provided, subject to availability of central office facilities and telephone numbers in the serving central office. Customers utilizing DID must arrange to provide for interception of all unused telephone numbers in the groups of DID telephone numbers assigned by the Telephone Company.

#### B. Rates and Charges

The following rates and charges are in addition to rates and charges applicable for PBX trunks:

		Nonrecurring		Mont	<u>hly</u>
		Minimum	Maximum	Minimum	Maximum
Des	cription /Billing Code/	Charge	Charge	Charge	Charge
1.	Each group of 20 DID or fraction thereof /ND4/ Or	numbers \$100.00 (2)	\$225.00	\$1.20	\$5.20
	Each individual DID DID number /SOT/	-	-	.10	.50
2.	Each DID trunk termination in central	115.00 office /NDT/	250.00	14.50	25.00

- 3. The customer shall provide all switching or common equipment located on his premises which is necessary for the provision of DID service.
- 4. Special intercept on DID numbers will be furnished under the regulations and rates set forth in the Ameritech Catalog.
- (1) Current prices for these services are shown in Section 12.11 of this tariff.
- (2) In lieu of the nonrecurring charge included in paragraph B.1 above, the nonrecurring charge per additional group of 20 DID numbers provided on the same occasion as an initial group of 20 DID numbers is set at a minimum charge of \$45.00 and a maximum charge of \$70.00. The current price is shown in Section 12.11 of this tariff.

Issued:	June 25, 2002	Effective: June 25, 2002
In accor	dance with Case No. 00-21-TP-ACE, issuedJune 20, 2002	·
Issued b	y: Louis F. McAlister, Jr., President	
	Navigator Telecommunications, LLC.	
	8525 Riverwood Park Drive, P.O. Box 13860	)
	North Little Rock, AR 72113-0860	

### 3.11 Central Office Services, cont. (1)

3.11.2 Two-Way Direct Inward Dialing (DID) Service With Call Transfer, cont.

#### B. Regulations, cont.

- 7. A change in CO equipment could require the customer to discontinue the service or obtain service from another CO. The Company makes no guarantees and assumes no liability for loss of service to the customer, resulting from the conversion or upgrade of CO equipment.
- 8. Two-Way DID With Call Transfer works in conjunction with DID Service and the rates and charges specified in Paragraph C following are in addition to the rates and charges for DID Service specified in Paragraph 1.B of this section. Local and Message Toll charges are applicable.

## C. Rates and Charges

Service Establishment

	Non	recurring	Mor	<u>ithly</u>
	Minimum	Maximum	Minimum	Maximum
Description /Billing Code/	Charge	Charge	Charge	Charge
1. With initial DID Service, per Trunk Group /SEPD1/	\$125.00 '	\$225.00	-	-
2.Subsequent to Establishme of DID service, per Trunk Group /SEPD1/		225.00	-	-
3.Two-Way DID Trunk Wit Call Transfer /3CW/ Conversions	h -	-	18.00	26.00
1.Of entire DID Trunk Grou to Two-Way DID or entire Two-Way DID Trunk Gro To DID /NR9DD/	e	375.00	• .	-
2.Of Individual DID Trunks to a new Two-Way DID Trunk Group, per Trunk Group /SEPD1/	125.00	225.00	-	-

Issued: June 25, 2002			Effective: June 25, 2002
In accordance with Case No.	00-21-TP-ACE, issued _	June 20, 2002	,
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommun	ications, LLC.	
85:	25 Riverwood Park Drive	, P.O. Box 13860	
	North Little Rock, AR	72113-0860	

### 3.11 Central Office Services, cont.

3.11.2 Two-Way Direct Inward Dialing (DID) Service With Call Transfer, cont.

C. Rates and Charges, cont.

\*\*

Conversions, cont.

	Non	recurring_	Mor	<u>nthly</u>
	Minimum	Maximum	Minimum	Maximum
Description /Billing Code/	Charge	Charge	Charge	Charge
3.Two-Way DID Trunk Wit	:h -	-	\$18.00	\$28.00
with Call Transfer /3CW/				
4. Change in Outpulsing, Start Dial, or Signal Type, per Trunk Group /REAJA		46.55	-	-
5.Change or Redesign in Signaling or Transmission Interface, per occurrence / REAJB/		175.00	-	-

## 3.12 Calling Party Number (CPN) Blocking (1), (2), (3)

Non-Residence

3.12.1 Per Line Calling Party Number Blocking (Calling Number Delivery Suppression)

The following rates and charges are applicable to Per Line Calling Party Number Blocking. See Section 2.6.2 for additional information related to this service.

	Nonrecurring		
	Minimum	Maximum	
Description	Charge	Charge	
Per Exchange Service,	\$6.30	\$12.30	
Residence/Non-Residence/No-Residence	lence		
	Mont	hly	
		<u>hly</u> Maximum	
Description			
Description Per Exchange Service	Minimum	Maximum	

(1) Current prices for these services are shown in Section 12.1 of this tariff.

(2) The monthly rate for Blocking will not exceed the monthly rate for private listing service.

.50

1.50

(3) Per Line Calling Party Number Blocking will be provided at no charge to qualified social service agencies, law enforcement organizations, and their certified employees, and volunteers. Per Line Calling Party Number Blocking will also be provided at no charge to customer-owned coin-operated telephone (COCOT) customers.

Issued: June 25, 2002 Effective: June 25, 2002 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

#### 3.13 SmartATM

### 3.13.1 Description

- A. The SmartATM Service is only available to business subscribers and provides the following services (includes Touch-Tone):
  - 1. Business individual lines for Automated Teller Machines only.
  - 2. Unlimited calling to 800 toll-free numbers.
- B. The rate specified herein is subject to a 12-month service term.
- 3.13.2 Regulations and Service Limitations
  - A. Calls made outside the service description shown in Section 3.13.1 are subject to normal usage charges as described in other sections of this tariff.
  - B. Charges for this service commence when the company's information records are posted and are payable monthly in advance.
  - C. Charges for service are automatically discontinued upon service termination.
  - D. If service is terminated before the 12-month term, a charge of \$10 (no minimum and maximum pricing) per month will be applied for each line currently or previously classified as a SmartATM line. This charge will be imposed based upon the remainder of the commitment period.
  - E. Existing SmartATM subscribers may not take advantage of special promotions or rate changes prior to the end of the 12-month service term. This is applicable to existing lines only. New lines may qualify for any special promotions or price changes.

...

# SECTION 3 - RATES - LOCAL EXCHANGE SERVICES, CONT.

## 3.13 SmartATM, cont.

3.13.3 Rates and Charges (1)

A. The following monthly rates (minimum/maximum) will apply for the SmartATM service.

Service Category	Rate Element	Minimum	Maximum
SmartATM	Access Area B	\$ 9.00	\$27.00
	Access Area C	10.00	30.00
	Access Area D	11.00	33.00

B. Message rate, per minute

Service Category	Rate Element	Minimum	Maximum
Per minute	All Access Areas	\$0.025	\$0.075

C. Installation Charge, per line

Service Category	Rate Element	Minimum	Maximum
Installation	All Access Areas	\$42.50	\$94.28
Charge, per line			

(1) See Section 2.3 for Access Area Designations.

Issued: June 2, 2006	Effective: June 5, 2006
In accordance with Case No. 00-21-TP-ACE	e, issued <u>June 20, 2002</u> .
Issued by: Louis F. Mc	Alister, Jr., President
Navigator Tel	ecommunications, LLC.
8525 Riverwood F	Park Drive, P.O. Box 13860
North Little	Rock, AR 72113-0860

#### 3.14 Business - Bundled Service Offerings (1)

- 3.14.1 SmartTime Unlimited for Business
  - A. Description
    - 1. This offering is available only to business customers (coin lines are not eligible) and provides for unlimited local and long distance calling. The customer may choose an unlimited number of features as shown in Section 3.9 of this tariff. Touch tone calling is included as part of this offering.
  - B. Regulations
    - 1. The customer must commit to at least a one-year term agreement to be eligible for the service.
    - 2. Maximum of ten (10) lines per location. Charges below are shown on a per line basis.
    - 3. Includes unlimited long distance calls for business customers, nondata, voice use only, within the 48 contiguous United States. Calls made to points outside of the contiguous 48 states, including all international calls, are not included in this offer, and will be billed at applicable rates. Subsequent to customer notification, Navigator may cancel subscriptions to any customer at any time due to fraud or misuse. Usage can be screened for other than voice usage through a review of call records for those customers identified as having significant usage. Taxes, fees, and any EAS charges also apply.
  - C. Rates and Charges
    - 1. SmartTime Unlimited for Business Recurring Charge, per line

	Minimum	Maximum
All Access Areas - Monthly - Each	\$35.74	\$74.25

2. SmartTime Unlimited for Business- Nonrecurring Charge, per line

	Minimum	Maximum
Phone Line Installation Fee	\$56.56	\$84.85

(1) See Section 2.3 for Access Area Designations.

Issued:	June 2, 2006	Effective: June 5, 2006
In accor	rdance with Case No. 00-21-TP-ACE, issued	June 20, 2002
Issued b	by: Louis F. McAlister, Jr., P	President
	Navigator Telecommunica	ations, LLC.
	8525 Riverwood Park Drive, P	.O. Box 13860
	North Little Rock, AR 72	113-0860

...

#### SECTION 3 - RATES - LOCAL EXCHANGE SERVICES, CONT.

#### 3.14 Business - Bundled Service Offerings (1), cont.

- 3.14.2 SmartTime for Business
  - A. Description
    - 1. This offering is available only to business customers (coin lines are not eligible) and provides for unlimited local calling. Long distance calling is billed at a minimum of \$.049 to a maximum of \$0.07 per minute. The customer may choose an unlimited number of features as shown in Section 3.9 of this tariff. Touch tone calling is included as part of this offering.

#### B. Regulations

- 1. The customer must commit to at least a one-year term agreement to be eligible for the service.
- 2. Long distance calls for business customers at \$.055 per minute is only for calls within the 48 contiguous United States. Calls made to points outside of the contiguous 48 states, including all international calls, are not included in this offer, and will be billed at applicable rates. Taxes, fees, and any EAS charges also apply.
- C. Rates and Charges
  - I. SmartTime for Business Recurring Charge, per line

	Minimum	Maximum
All Access Areas - Monthly - Each	\$29.99	\$44.55

2. SmartTime for Business – Nonrecurring Charge, per line

	Minimum	Maximum
Phone Line Installation Fee	\$56.56	\$84.85

(1) See Section 2.3 for Access Area Designations.

## SECTION 4 - PUBLIC TELEPHONE SERVICE, CONT.

#### 4.1 Independent Payphone Provider Services, cont.

- 1. Customer-Owned, Coin Operated Telephones (COCOT), cont.
  - E. Prices, cont.
    - 1. Service Elements, cont.

d. Answer Supervision – Line S	Side	
-	Minimum Nonrecurring	Maximum Nonrecurring
Description	Charge	Charge
Answer Supervision equipped with line side interface, per line		
<ul> <li>nonrecurring charge</li> </ul>	\$.50	\$7.00
- monthly price	\$.60	\$5.20

2. Other Applicable Charges and Payments

In addition to Telephone Exchange Service Network Access Rates, including Network Access Line Quantity Discount Rates, end user access line charges shown in Navigator's Telecommunications, LLC.'s Tariff F.C.C. No. 3 are also applicable.

Issued: June 25, 2002		-	Effective: June 25, 2002
In accordance with Case N	lo. 00-21-TP-ACE, issued	June 20, 2002	•
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommun	ications, LLC.	
	8525 Riverwood Park Drive	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

## SECTION 4 – PUBLIC TELEPHONE SERVICE, CONT.

## 4.1 Independent Payphone Provider Services, cont.

2. COCOT - Coin Line, cont.

#### D. Prices

#### 1. Service Elements

Description	Access Area	Minimum Monthly Charge	Maximum Monthly Charge
Telephone Exchange Service Network Access Rates			
Network Access, per Coin Line Network Access, per Coin Line Network Access, per Coin Line Network Access, per Coin Line	A B C D	- \$23.45 \$25.45 \$27.95	- \$27.45 \$29.45 \$31.95

2. Other Applicable Charges and Payments

In addition to the above rates, interstate End User Common Line (EUCL) charges are also applicable.

Issued: June 25, 2002			Effective: June 25, 2002
In accordance with Case	No. 00-21-TP-ACE, issued	June 20, 2002	
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommun	ications, LLC.	
	8525 Riverwood Park Drive,	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

...

## SECTION 4 - PUBLIC TELEPHONE SERVICE, CONT.

#### 4.1 Independent Payphone Provider Services, cont.

- 2. COCOT Coin Line, cont.
  - D. Prices, cont.
    - 3. Other Optional Offerings

### **Outgoing Only Service**

An optional service offered to customers who wish to maximize payphone usage or reduce the ability of end users to conduct business from a location that prohibits incoming calls. Callers hear a recording stating that the phone is not in service for incoming calls. It is the IPP's responsibility to state this restriction on the instruction card of the phone.

## **Restricted Coin Access (RCA)**

An optional service for public telephones where a coin collecting device can be disabled during certain hours at the IPP's discretion to permit origination of Local calls, extended Local calls and Message Toll calls to only be charged to a third number, a Company calling card, or placed on a collect basis. Calls to public emergency numbers such as 911 will continue to be permitted and provided free of charge. End users placing coin calls during restricted hours will have their coins returned and will hear a recording that the phone is not equipped for coin calls during certain hours. It is the IPP's responsibility to state this restriction on the instruction card of the phone.

Decemination	Minimum Nonrecurring	Maximum Nonrecurring
Description	Charge	Charge
Outgoing Only Service		-
Restricted Coin Access	\$200.00	\$230.00
	Minimum	Maximum
	Monthly	Monthly
Description	Charge	Charge
Outgoing Only Service	-	-
Restricted Coin Access	\$8.75	\$15.75

Issued: June 25, 2002	· · · · · · · · · · · · · · · · · · ·		Effective: June 25, 2002
In accordance with Case	No. 00-21-TP-ACE, issued	June 20, 2002	<u>     .</u> .
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommun	ications, LLC.	
	8525 Riverwood Park Drive,	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

## SECTION 6 - RULES AND REGULATIONS – INTEREXCHANGE SERVICES, CONT.

#### 6.7 Billing and Payment for Service, cont.

6.7.3 If a Customer is not satisfied with the Company's response to an inquiry or request for credit, he or she may appeal to the Public Utilities Commission of Ohio for final resolution. The Public Utilities Commission of Ohio may be contacted at:

Public Utilities Commission of Ohio 180 E. Broad Street Columbus, OH 43215-3793 Toll-free telephone: 1-800-686-PUCO (7826)

#### 6.7.4 Delinquent Bills

Navigator Telecommunications, LLC. allows residential Customers at least 21 days from the date of the postmark of the bill to pay bill charges.

#### 6.7.5 Late Payment Fees

The Company reserves the right to assess a one-time late payment fee of 1.5% per month on any past due balance. The late payment charge will not be applied to previous late payment charges that have been assessed but not yet been paid for, but will apply to the accumulated services for which the customer is in arrears. Late payment charges will be applied without discrimination.

6.7.6 Returned Check Charge

There is a minimum charge of \$15 and a maximum charge of \$35 for each check returned for insufficient funds. The Company will charge \$20 for each check returned for insufficient funds. This charge may be waived under appropriate circumstances.

# SECTION 8 - MISCELLANEOUS SERVICES AND RATES – INTEREXCHANGE SERVICES

## 8.1 Directory Assistance

Directory Assistance is available to Customers of Navigator Telecommunications, LLC. Directory Assistance charges apply to each call to the Directory Assistance Bureau. Up to two requests may be made on each call to Directory Assistance. The Directory Assistance charge applies to each call regardless of whether the Directory Assistance Bureau is able to furnish the requested telephone number.

	Minimum	Maximum
	Per Use	Per Use
	<u>Charge</u>	<u>Charge</u>
Directory Assistance	\$.45	\$1.75

(1) Current prices for these services are shown in Section 12.14 of this tariff.

#### **SECTION 12 – PRICE LIST**

#### 12.1 General

Rates shown in this section of the tariff are the actual prices currently being charged by Navigator Telecommunications, LLC. for the services shown. Minimum and maximum rates for these services are referenced in this section and are shown in other sections of this tariff.

## 12.2 Service Charges (1)

12.2.2 Nonrecurring Charges

A. Application of Nonrecurring Charges to Establish Services

Except where otherwise specified in this paragraph, the same nonrecurring charges apply to residence and non-residence service.

The Central Office Connection and Line Connection Charges do not apply to the transfer of an existing central office line servicing a reseller's end user when there is no interruption of service.

1. Exchange Service

The following charges are current rates applicable to each individual exchange service being established:

-	Complex		Sin	nple
		Non-		Non-
	<u>Residence</u>	Residence	<b>Residence</b>	<u>Residence</u>
a. Local Exchange Service				
- Service Ordering				
Charge, per location,				
per occasion	\$17.65	\$15.85	\$17.65	\$25.50
- Central Office				
Connection Charge,				
per termination	8.25	17.00	8.25	13.00
- Line connection charg	e,			
per termination	10.60	16.50	10.60	24.35

(1) Min/max ranges for these services are shown in Section 3.2 of this tariff.

Issued: June 25, 2002			Effective: June 25, 2002
In accordance with Cas	e No. 00-21-TP-ACE, issued	June 20, 2002	
Issued by:	Louis F. McAlister, Jr.,	, President	
	Navigator Telecommuni	cations, LLC.	
	8525 Riverwood Park Drive,	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

## 12.2 Service Charges, cont. (1)

## 12.2.2 Nonrecurring Charges

- A. Application of Nonrecurring Charges to Establish Service, cont.
  - 1. Exchange Service, cont.

The following charges are current rates applicable to each individual exchange service being established:

	Com	plex	Sin	nple
_		Non-		Non-
Re	esidence	<b>Residence</b>	<u>Residence</u>	<b>Residence</b>
b Central Office				
Connection Charge,				
per termination	-	-	-	13.00
- Line Connection Charge,				
per termination (includes	·			
installation of telephone)	-	-	-	63.00
c. Centrex stations and termin	nals			
Exhibition Hall stations, A				
Switching system stations	-			
- Service Ordering Charge				
per location, per occasion	-	46.80	_	_
- Central Office Connectio				
Charge, per station				
termination	-	1.75	-	-
- Line Connection Charge				
per station termination	-	3.35	-	-
-				
d. PBX Trunks				
- Service Ordering Charge				
per location, per occasion		24.50	-	-
- Central Office Connectio	n			
Charge, per termination	14.50	14.50	-	-
- Line Connection Charge,				
per termination	10.35	10.35	-	-

(1) Min/max ranges for these services are shown in Section 3.2 of this tariff.

Issued: June 25, 2002			Effective: June 25, 2002
In accordance with Case No.	00-21-TP-ACE, issued	June 20, 2002	<u></u> .
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommuni	cations, LLC.	
852	25 Riverwood Park Drive,	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

#### **12.2** Service Charges, cont. (1)

12.2.2 Nonrecurring Charges

- A. Application of Nonrecurring Charges to Establish Service, cont.
  - 1. Exchange Service, cont.

The following charges are current rates applicable to each individual exchange service being established:

	Com	Complex		Simple	
		Non-		Non-	
	Residence	Residence	Residence	Residence	
e.	Misc. Services/Features (2)				
	- Service Ordering Charge				
	per location, per occasion 7.95	7.30	6.00	7.30	

- (1) If required, min/max ranges for these services are shown in Section 3.2 of this tariff
- (2) Miscellanous Services and Features charges apply per individual exchange service, per occasion to Custom Calling Service and Advanced Custom Calling Service (other than Calling Party Number Blocking or Call Trace) excluding Call Waiting and Caller ID.

Issued: October 5, 2006 Effective: October 6, 2006 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 \*\*

# SECTION 12 – PRICE LIST, CONT.

# 12.2 Service Charges, cont. (1)

# 12.2.2 Nonrecurring Charges

A. Application of Nonrecurring Charges to Establish Service, cont.

2. Network Wiring Charge

- · ·		Nonrecurring
		Charge
-	Per semi-public telephone	\$86.50
-	Change to/from handset with	75.55
	volume control	

# B. Application of Nonrecurring Charges for Restoral of Service

	<u>Complex</u>		Sin	<u>nple</u> Non-
	Residence	Residence	Residence	Residence
1. Service restored after temporary denial but prior to completion of order to discontinue service, per service or system				
Restoral Charge	\$26.55	\$26.55	\$32.30	\$32.30
2. Service restored after temporary suspension				
Restoral Charge	62.30	75.85	33.55	33.55
3. Service restored after temporary interception				
Restoral Charge	62.50	87.70	33.55	33.55

(1) Min/max ranges for these services are shown in Section 3.2 of this tariff.

Issued:	June 25, 2002	Effective: June 25, 2002
In accord	lance with Case No. 00-21-TP-ACE, issuedJune 20, 2002	<u> </u>
Issued by	Louis F. McAlister, Jr., President	
	Navigator Telecommunications, LLC.	
	8525 Riverwood Park Drive, P.O. Box 13860	
	North Little Rock, AR 72113-0860	

# 12.3 Local Service Rates (1)

# 12.3.1 Flat Rate Service

Flat Rate service consists of a usage package that includes an unlimited number of local messages.

A. Rates and Charges

	Usage Package Monthly Rate
Residence PBX Trunks	\$ 8.95
Residence Individual	\$ 7.55

### 12.3.2 Local Service Rates

# A. Schedule of Monthly Rates – Network Access Lines

Description	Access Area	Network Access Line	Central Office Termination	Total Monthly <u>Rate</u>	
Non - Residence					
- Non-Rotary	А	Future Use	-	-	
With Touch-Tone	В	\$15.95	\$2.30	\$18.25	•
	С	17.95	2.30	20.25	
	D	20.45	2.30	22.75	
- Rotary or	А	Future Use	-	-	
- PBX Trunk	В	\$15.95	\$6.00	\$21.95	
With Touch-Tone	С	17.95	6.00	23.95	
	D	20.45	6.00	26.45	

(1) Min/max ranges for these services are shown in Section 3.3 of this tariff.

Issued: June 25, 2002			Effective: June 25, 2002
In accordance with Case N	o. 00-21-TP-ACE, issued	June 20, 2002	<u> </u>
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommun	ications, LLC.	
8	525 Riverwood Park Drive	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

-

# **SECTION 12 – PRICE LIST, CONT.**

# 12.3 Local Service Rates, cont. (1)

# 12.3.2 Local Service Rates, cont.

# A. Schedule of Monthly Rates – Network Access Lines, cont.

Description	Access Area	Network Access Line	Central Office Termination	Total Monthly Rate
Non - Residence				
- Non-Rotary	А	Future Use	-	-
Without	В	\$15.95	\$2.30	\$18.25
Touch-Tone	С	17.70	2.30	20.00
	D	17.70	2.30	20.00
- Rotary or	А	Future Use	-	-
- PBX Trunk	В	\$15.95	\$6.00	\$21.95
Without	С	17.70	6.00	23.70
Touch-Tone	D	17.70	6.00	23.70
Residence				
- Individual	А	Future Use	-	-
With Touch-Tone	В	\$4.40	\$2.30	\$6.70
	С	4.40	2.30	6.70
	D	4.40	2.30	6.70
- PBX Trunk	А	Future Use	-	-
With Touch-Tone	В	\$15.95	\$6.00	\$21.95
	С	17.95	6.00	23.95
	D	20.45	6.00	26.45

(1) Min/max ranges for these services are shown in Section 3.3 of this tariff.

Issued: June 25, 2002			Effective: June 25, 2002
In accordance with Cas	se No. 00-21-TP-ACE, issued	June 20, 2002	
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommuni	cations, LLC.	
	8525 Riverwood Park Drive,	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

...

# 12.3 Local Service Rates, cont. (1)

12.3.2 Local Service Rates, cont.

# A. Schedule of Monthly Rates – Network Access Lines, cont.

Description	Access Area	Network Access Line	Central Office Termination	Total Monthly Rate
Residence				
- Individual	А	Future Use	-	-
Without	В	\$4.40	\$2.30	\$6.70
Touch-Tone	С	4.40	2.30	6.70
	D	4.40	2.30	6.70
- PBX Trunk	A	Future Use	-	-
Without	В	\$15.95	\$6.00	\$21.95
Touch-Tone	С	17.70	6.00	23.70
	D	17.70	6.00	23.70

## 12.3.3 Measured Rate Services

A. Local Message Charge Schedule

	Initial Minute	Additional Minute
Rate Mileage	or Fraction Thereof	or Fraction Thereof
0 - 10	\$.0353	\$.0088
11 - 22	.0397	.0132
23 and over	.0442	.0177

(1) Min/max ranges for these services are shown in Section 3.3 of this tariff.

Issued: June 25, 2002		Effective: June 25, 2002
In accordance with Case	No. 00-21-TP-ACE, issued June 20, 2002	<u> </u> .
Issued by:	Louis F. McAlister, Jr., President	
	Navigator Telecommunications, LLC.	
	8525 Riverwood Park Drive, P.O. Box 1386	0
	North Little Rock, AR 72113-0860	

\*\*

# SECTION 12 - PRICE LIST, CONT.

#### 12.3 Local Service Rates, cont. (1)

### 12.3.4 Local Calling Plus

#### A. Local Calling Plus Charge Schedule

	Initial Minute	Additional Minute
Rate Mileage	or Fraction Thereof	or Fraction Thereof
0 - 10	\$.0390	\$.01
11 - 22	.0430	.015
23 and over	.0490	.02

(1) Min/max ranges for these services are shown in Section 3.3 of this tariff.

 Issued:
 June 25, 2002
 Effective:
 June 25, 2002

 In accordance with Case No. 00-21-TP-ACE, issued
 June 20, 2002
 .

 Issued by:
 Louis F. McAlister, Jr., President

 Navigator Telecommunications, LLC.
 8525 Riverwood Park Drive, P.O. Box 13860

 North Little Rock, AR 72113-0860

••

# SECTION 12 – PRICE LIST, CONT.

# 12.4 Operator Service Rates – Traditional (1)

12.4.1 Service Charges

12,7,1	Service Charges		
		Per Use	
		Charge	
	Automated Calling Card	\$.95	(I)
	Station-to-Station		
	Customer Dialed-Operator	1.95	
	Assisted-Calling Card		
	Station-to-Station		1
	Operator Assisted	1.75	
	Station-to-Station		
	Operator Assisted	3.75	
	Person-to-Person		
	Operator Assisted	2.25	
	Third Number Billed		
12.4.2	Directory Assistance Service - Local		
	Residence Charge		
	Where customer direct dials	\$1.10	
	directory assistance number		
	Where customer requests operator assis	tance to place a	(C)
	call to directory assistance, the surcha	-	Ĩ
	Non - Residence Charge		
	Where customer direct dials	1.10	
	directory assistance number		
	Where customer requests operator assis	stance to place a	(C)
	call to directory assistance, the surcha	-	
	National Directory Assistance		
	Direct Dialed	1.50	(I)
	Alternate Billed Calls	1.50	Ĩ
(1) Min/m	nax ranges for these services are shown in	Section 3.4 of this tariff.	

Issued: April 23, 2004			Effective: April 26, 2004
In accordance with Case N	o. 00-21-TP-ACE, issued	June 20, 2002	•
Issued by:	Louis F. McAlister, Jr.	., President	
	Navigator Telecommun	ications, LLC.	
5	3525 Riverwood Park Drive	, P.O. Box 13860	)
	North Little Rock, AR	72113-0860	

...

# SECTION 12 – PRICE LIST, CONT.

# 12.4 Operator Service Rates – Traditional, cont. (1)

12.4.3 Information Call Completion (ICC) Service

	Per Use
	Charge
ICC, per call	.25

# 12.4.4 Busy Line Verify and Line Interrupt Service

	Per Use	
	Charge	
Verification	1.75	(I)
Interrupt	2.00	I

(1) Min/max ranges for these services are shown in Section 3.4 of this tariff.

Issued: April 23, 2004			Effective: April 26, 2004
In accordance with Case No. 00-	21-TP-ACE, issued	June 20, 2002	·
Issued by:	Louis F. McAlister, Jr.,	President	
Ν	avigator Telecommuni	cations, LLC.	
8525 F	Riverwood Park Drive,	P.O. Box 13860	
1	North Little Rock, AR (	72113-0860	

٠.

# SECTION 12 - PRICE LIST, CONT.

# 12.5 Alternative Operator Service Rates (1)

# 12.5.1 Prices

1. Operator Assisted Services

Description	Nonrecurring Charge
Automated Calling Card Station-to-Station, Per call	\$0.70
Customer-Dialed Calling Card Station-to-Station, per call	1.70
Operator-Assisted Station-to-Station, Per call	2.50
Operator-Assisted Person-to-Person, Per call	4.80
Operator-Assisted Third Number Billed, Per call	2.50

(1) Min/max ranges for these services are shown in Section 3.5 of this tariff.

Issued: June 25, 2002	· · · · · · · · · · · · · · · · · · ·		Effective: June 25, 2002
In accordance with Case No. 00	)-21-TP-ACE, issued	June 20, 2002	•
Issued by:	Louis F. McAlister, Jr.	, President	
[	Navigator Telecommuni	ications, LLC.	
8525	Riverwood Park Drive,	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

#### 12.5 Alternative Operator Service Rates, cont. (1)

#### 12.5.1 Prices

#### 2. Message Toll Service Rates (2), per minute

	Day ]	Rates	Evening	<u>g Rates</u>	Night & Wo	eekend Rates
Rate	Initial	Add'l	Initial	Add'l	Initial	Add'l
Mileage	Minute	Minute	Minute	Minute	Minute	<u>Minute</u>
Schedule A (3)						
1-10	\$.23	\$.23	\$.16	\$.16	\$.16	<b>\$</b> .16
11-22	.23	.23	.16	.16	.16	.16
22-55	.23	.23	.16	.16	.16	.16
56-124	.23	.23	.16	.16	.16	.16
125-End	.23	.23	.16	.16	.16	.16
Schedule B (3)						
1-10	\$.23	\$.23	\$.16	\$.16	\$.16	<b>\$.16</b>
11-22	.23	.23	.16	.16	.16	.16
22-55	.23	.23	.16	.16	.16	.16
56-124	.23	.23	.16	.16	.16	.16
125-End	.23	.23	.16	.16	.16	.13
Operator Assisted Ca	<u>lls – Inmate</u>	Services (4)				
1-10	\$.25	\$.16	\$.12	\$.06	\$.12	\$.06
11-22	.28	.16	.160	.10	.160	.10
22-55	.28	.20	.160	.10	.160	.10
56-124	.28	.20	.160	.10	.160	.10
125-End	.28	.20	.160	.10	.160	.10

(1) Min/max per minute ranges for these services are shown in Section 3.5 of this tariff.

- (2) Day Rates Monday thru Friday 8:00 AM to but not including 5:00 PM Evening Rates – Monday thru Friday 5:00 PM to but not including 11:00 PM Night & Weekend Rates – Monday thru Friday 11:00 PM to but not including 8:00 AM Saturday, Sunday, and Holidays – All Day
- (3) Schedule A rates apply to calls terminating/originating in an Independent Telephone Company Exchange. Schedule B rates apply for customer-dialed station-to-station calling.
- (4) Maximum rates are as shown in the PUCO's Service Requirements Form.

#### 12.5 Alternative Operator Service Rates, cont. (1)

#### 12.5.1 Prices

...

2. Message Toll Service Rates, cont.

MTS Rate Discounts

Description	Discount
\$ 0.00 - \$25.00	0%
25.00 - 100.00	10%
100.01 - 200.00	15%
200.01 and up	20%

(1) Min/max ranges for these services are shown in Section 3.5 of this tariff.

 Issued:
 June 25, 2002
 Effective:
 June 25, 2002

 In accordance with Case No. 00-21-TP-ACE, issued
 June 20, 2002
 .

 Issued by:
 Louis F. McAlister, Jr., President

 Navigator Telecommunications, LLC.
 8525 Riverwood Park Drive, P.O. Box 13860

 North Little Rock, AR 72113-0860

## 12.6 Directory Services

**(T)** 

## 12.6.1 Directory Listings

The following monthly rates apply to both regular and special types of additional listings:

	Monthly		
Non-residence, additional listing	<u>Charge</u> \$3.00	USOC CLT	(1)
Non-residence, foreign listing	3.00	FLT	(N) 
Residence, additional listing	.90	RLT	

#### 12.6.2 Private and Semi-Private Listing Service

Private listing service, each service	Monthly <u>Charge</u> 2.10	<u>USOC</u> NPU	(1)
Semi-private listing service, each service	2.10	NLT	(I)(T)
Optional arrangement for relaying messages to private listing service customers, each message	1.80	-	

(D)

Issued: October 5, 2006 Effective: October 6, 2006 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

## 12.7 Call Blocking (1)

# 12.7.1 Rates and Charges

1. Residence and Non-Residence Customer Requested Call Blocking

The following rates and charges will be waived for the initial establishment of call blocking. These charges will apply to subsequent requests for blocking.

...

	Nonrecurring Charge	<u>USOC</u>
a. Residence Service		
Call blocking, per service	\$ 9.30	RTVXN
b. Non-Residence Service		
Call blocking, per individual		
or trunk or WATS service	\$ 9.30	RTVXN
c. Centrex Service		
Call blocking, Centrex stations	\$41.55	RTVXN
2. Sponsor Requested Call Blocking		
The following rates and charges apply	y to sponsor requested	call blocking:
a. Residence Service		
Call blocking, per request,		
per service	\$60.00	NR9CB
b. Non-Residence Service		
Call blocking per request, per		
individual or trunk		
or WATS service	\$60.00	NR9CB
c. Centrex Service		
Call blocking, per request, per		
Centrex system	\$60.00	NR9CB

(1) Min/max ranges for these services are shown in Section 3.7 of this tariff.

Issued:	June 25, 2002	Effective: June 25, 2002
In accor	rdance with Case No. 00-21-TP-ACE, issuedJune 20, 2002	<b>*</b>
Issued b	by: Louis F. McAlister, Jr., President	
	Navigator Telecommunications, LLC.	
	8525 Riverwood Park Drive, P.O. Box 13860	
	North Little Rock, AR 72113-0860	

\*\*

## SECTION 12 – PRICE LIST, CONT.

#### 12.8 Toll Restriction (1)

#### 12.8.1 Rates and Charges

The following rates and charges apply to toll restriction service and are in addition to all other rates and charges applicable to residence, non-residence, or PBX trunk service. See Section 3.4 for local service rates for these services.

Toll Restriction, per individual residence service	Nonrecurring <u>Charge</u> \$12.00	<u>USOC</u> RTVX5
Toll Restriction, per individual non-residence service or PBX trunk equipped	\$24.40	KXT
Monthly Charges Toll Restriction, per individual residence service	Monthly <u>Rate</u> \$3.00	<u>USOC</u> RTVX5
Toll Restriction, per individual non-residence service or PBX trunk equipped	\$51.70	КХТ

(1) Min/max ranges for these services are shown in Section 3.8 of this tariff.

Note: A toll restriction charge will not apply to those residential customers: 1) who elect toll restriction in lieu of a deposit, 2) who elect toll restriction while making payments for a final bill, or 3) who elect payment arrangements and toll restriction in lieu of denial.

Issued: June 25, 2002 Effective: June 25, 2002 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 . Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 ų.

# **SECTION 12 – PRICE LIST, CONT.**

# 12.9 Custom Calling Features (1)

12.9.1 Residence - Rates and Charges

	Monthly Charge	
Description /Billing Code/	Per Line	
Call Waiting /ESX/	\$4.15	
Call Forwarding /ESM/	4.00	
Call Forwarding /TAS/CFW/	4.00	
Three-Way Calling /ESC/	4.00	
Description /Billing Code/	Per Use Charge	
Three-Way Calling	\$.70	(T)

# 12.9.2 Business - Rates and Charges

	Monthly Charge	
Description /Billing Code/	Per Line	
Call Waiting /ESX/	\$6.00	(I)
Call Forwarding /ESM/	7.00	
Call Forwarding /TAS/CFW/	6.50	
Three-Way Calling /ESC/	6.50	1
Speed Calling 8/ESL	6.00	(N)
Speed Calling 30/ESF	7.00	ļ
Description /Billing Code/	Per Use Charge	
Three-Way Calling	\$1.99	(T)(I)

(1) Where required, min/max ranges for these services are shown in Section 3.9 of this tariff. (T)

Issued: October 5, 2006		Effective:	October 6, 2006
In accordance with Case	No. 00-21-TP-ACE, issued	June 20, 2002	
Issued by:	Louis F. McAlister, Jr.,	President	
	Navigator Telecommuni	cations, LLC.	
	8525 Riverwood Park Drive,	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

-

# **SECTION 12 – PRICE LIST, CONT.**

# 12.10 Advanced Custom Calling Features (1)

# 12.10.1 Residence - Rates and Charges

	Monthly Charge	
Description /Billing Code/	Per Line	
Repeat Dialing /NSS/	\$4.10	
Caller ID /NSD/	6.00	
Caller ID w/Name /NMP/ (See Note)	1.95	
Automatic Callback /NSQ/	4.10	
Call Screening /NSY/	4.00	
Description	Per Use Charge	
Repeat Dialing	\$.75	
Automatic Callback	.70	
Call Trace, per successful activation	3.50	
12.10.2 Non-Residence - Rates and Charges	Monthly Charge	
Description /Billing Code/	Per Line	
Repeat Dialing /NSS/	\$6.50	(I)
Caller ID /NSD/	7.00	(*)
Caller ID //NSD/ Caller ID w/Name /NMP/ (See Note)	4.20	(I)
Automatic Callback /NSQ/	6.00	(I) (I)
Call Screening /NSY/	4.00	(1)
Call Waiting ID/NWT/	2.00	(N)
Can waiting 1D/N w 17	2.00	(i)
Description	Per Use Charge	
Repeat Dialing	\$.75	
Automatic Callback	1.99	(I)
Call Trace, per successful activation	4.75	(I)

Note: The monthly rate for Caller ID with Name is in addition to the monthly rate for Caller ID.

(1) Where required, min/max ranges for these services are shown in Section 3.10 of this tariff. (T)

Issued: October 5, 2006		Effective: October 6, 2006
In accordance with Case N	o. 00-21-TP-ACE, issued	June 20, 2002
Issued by:	Louis F. McAlister, Jr.	, President
	Navigator Telecommuni	ications, LLC.
8	525 Riverwood Park Drive,	P.O. Box 13860
	North Little Rock, AR	72113-0860

#### **12.11** Central Office Services (1)

#### 12.11.1 Direct Inward Dialed (DID) Service

#### A. Rates and Charges

The following rates and charges are in addition to rates and charges applicable for PBX trunks:

		Nonrecurring	Monthly
Des	scription /Billing Code/	Charge	Charge
1.	Each group of 20 DID numbers or fraction	\$150.00 (2)	\$3.20
	thereof /ND4/		
	Or		
	Each individual DID number /SOT/	-	.25
2.	Each DID trunk termination in central	155.00	19.00
	office /NDT/		

- (1) Min/max ranges for these services are shown in Section 3.11 of this tariff.
- (2) In lieu of the nonrecurring charge included in Section 3.11.1.B.1, the nonrecurring charge per additional group of 20 DID numbers provided on the same occasion as an initial group of 20 DID numbers is \$54.20.

Issued: June 25, 2002 Effective: June 25, 2002 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

# 12.11 Central Office Services, cont. (1)

# 12.11.2 Two-Way Direct Inward Dialing (DID) Service With Call Transfer

# A. Rates and Charges

Service Establishment

\*\*

Service Establishment		
	Nonrecurring	-
Description/Billing Code/	Charge	Charge
1. With initial DID Service, per Trunk Group /SEPD1/	\$175.00	-
2.Subsequent to Establishment of	175.00	-
DID service, per Trunk Group /SEPD1/		
3.Two-Way DID Trunk With Call	-	23.00
Transfer / 3CW/		
Conversions		
1.Of entire DID Trunk Group to Two-Way DID or entire Two-Way DID Trunk Group to DID /NR9DD/	300.00	-
2.Of individual DID Trunks to a new Two-Way DID Trunk Group, per Trunk Group /SEPD1/	175.00	-
3.Two-Way DID Trunk with Call Transfer /3CW/	-	23.00
4. Change in Outpulsing, Start Dial, or Signal Type, per Trunk Group /REAJA/	41.55	-
5.Change or Redesign in Signaling or Transmission Interface, per occurrence /REAJB/	150.00	-

(1) Min/max ranges for these services are shown in Section 3.11 of this tariff.

Issued: Jun	e 25, 2002	Effective: June 25, 2002
In accordance	e with Case No. 00-21-TP-ACE, issuedJune 20, 2002	
Issued by:	Louis F. McAlister, Jr., President	
	Navigator Telecommunications, LLC.	
	8525 Riverwood Park Drive, P.O. Box 13860	
	North Little Rock, AR 72113-0860	

...

## SECTION 12 – PRICE LIST, CONT.

## 12.12 Calling Party Number (CPN) Blocking (1), (2), (3)

12.12.1 Per Line Calling Party Number Blocking (Calling Number Delivery Suppression)

The following rates and charges are applicable to Per Line Calling Party Number Blocking. See Section 2.6.2 for additional information related to this service.

Description	Nonrecurring Charge	Monthly Charge
Per Exchange Service - Residence	\$9.30	\$.50
<ul> <li>Non-Residence</li> </ul>	9.30	1.00

(1) Min/max ranges for these services are shown in Section 3.12 of this tariff.

..

## **SECTION 12 – PRICE LIST, CONT.**

#### 12.13 Independent Payphone Provider Services, cont. (1)

12.13.1 Customer-Owned, Coin Operated Telephones (COCOT), cont.

A. Prices, cont.

1. Service Elements, cont.

d. Answer Supervision – Line Side

	Nonrecurring	Monthly	
Description	Charge	Charge	
Answer Supervision equipped with line side			
Interface, per line	\$2.50	\$1.60	

2. Other Applicable Charges and Payments

In addition to Telephone Exchange Service Network Access Rates, including Network Access Line Quantity Discount Rates, end user access line charges shown in Navigator's Telecommunications, LLC.'s Tariff F.C.C. No. 3 are also applicable.

(1) Min/max ranges for these services are shown in Section 4.1 of this tariff.

Issued: June 25, 2002		, <u></u> _, <u>_</u> , <u>_</u>	Effective: June 25, 2002
In accordance with Case No.	00-21-TP-ACE, issued	June 20, 2002	<u> </u>
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommuni	cations, LLC.	
85	25 Riverwood Park Drive,	P.O. Box 13860	
	North Little Rock, AR	72113-0860	

# 12.13 Independent Payphone Provider Services, cont. (1)

2. COCOT - Coin Line

## A. Prices

## 1. Service Elements

Description	Access Area	Monthly <u>C</u> harge
Telephone Exchange Service Network Access Rates		
Network Access, per Coin Line	А	-
Network Access, per Coin Line	В	\$25.45
Network Access, per Coin Line	С	27.45
Network Access, per Coin Line	D	29.95

2. Other Applicable Charges and Payments

In addition to the above rates, interstate End User Common Line (EUCL) charges are also applicable.

(1) Min/max ranges for these services are shown in Section 4.1 of this tariff.

Issued: June 25, 2002			Effective: June 25, 2002
In accordance with Case N	lo. 00-21-TP-ACE, issued _	June 20, 2002	·
Issued by:	Louis F. McAlister, Jr.	, President	
	Navigator Telecommun	ications, LLC.	
:	3525 Riverwood Park Drive	, P.O. Box 13860	
	North Little Rock, AR	72113-0860	

14

## SECTION 12 - PRICE LIST, CONT.

# 12.13 Independent Payphone Provider Services, cont. (1)

2. COCOT - Coin Line, cont.

A. Prices, cont.

3. 0	Other	Optional	Offerings
------	-------	----------	-----------

	Nonrecurring	Monthly
Description	Charge	Charge
Outgoing Only Service	-	-
Restricted Coin Access (RCA)	\$212.00	\$10.75

(1) Min/max ranges for these services are shown in Section 4.1 of this tariff.

 Issued:
 June 25, 2002
 Effective: June 25, 2002

 In accordance with Case No. 00-21-TP-ACE, issued
 June 20, 2002
 .

 Issued by:
 Louis F. McAlister, Jr., President
 .

 Navigator Telecommunications, LLC.
 8525 Riverwood Park Drive, P.O. Box 13860
 .

 North Little Rock, AR 72113-0860
 .
 .

### 12.14 Directory Assistance – Interexchange Services

Directory Assistance is available to Customers of Navigator Telecommunications, LLC. Directory Assistance charges apply to each call to the Directory Assistance Bureau. Up to two requests may be made on each call to Directory Assistance. The Directory Assistance charge applies to each call regardless of whether the Directory Assistance Bureau is able to furnish the requested telephone number.

	Per Use	
	<u>Charge</u>	
Directory Assistance	\$1.25	(I)

(1) Min/max ranges for these services are shown in Section 8.1 of this tariff.

Issued: June 25, 2002 Effective: June 25, 2002 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

# Exhibit B

.

••

## CHECK SHEET

The Title Page and pages listed below of this tariff are effective as of the date shown. Revised sheets contain all changes from the original tariff that are in effect as of the date indicated.

PAGE	REVISION	PAGE	<u>REVISION</u>
Title Sheet	Original	31	Original
1	Fifth Revised *	32	Original
1.1	Second Revised *	33	Original
1.2	Fourth Revised *	34	Original
1.3	Fourth Revised *	35	Original
2	Original	36	Original
3	Original	37	Original
4	Second Revised	38	Oríginal
5	Original	39	Original
6	Second Revised	40	Oríginal
7	Original	41	Original
8	Original	42	Original
9	Original	43	Original
10	Original	44	Original
11	Original	45	Original
12	Original	46	Original
13	Original	47	Original
14	Original	48	Original
15	Original	49	Original
16	Original	50	Original
17	Original	51	Original
18	Original	52	Original
19	Original	53	Original
20	Original	54	Original
21	Original	55	Original
22	Original	56	Original
23	Original	57	Original
24	Original	58	Origina]
25	Original	59	Original
26	Original	60	Original
27	Original	61	Original
28	Original	62	Original
29	Original	63	Original
30	Original	64	Original

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860

North Little Rock, AR 72113-0860

,

## CHECK SHEET

••

The Title Page and pages listed below of this tariff are effective as of the date shown. Revised sheets contain all changes from the original tariff that are in effect as of the date indicated.

PAGE	<b>REVISION</b>	PAGE	<b>REVISION</b>
65	Original	99	Original
66	Original	100	Original
67	Original	101	Original
68	Original	102	Original
69	Original	103	Original
70	Original	104	Original
71	Original	105	Original
72	Original	106	Original
73	Original	107	Original
74	Original	108	Original
75	Original	109	Original
76	Original	110	Original
77	Original	111	Original
78	Original	112	Original
79	Original	113	Original
80	Original	114	Original
81	Original	115	Original
82	Original	116	First Revised
83	Original	117	Original
84	Original	118	Original
85	Original	119	Original
86	Original	120	First Revised *
87	Original	121	First Revised *
88	Original	122	Second Revised *
89	Original	123	First Revised *
90	Original	124	First Revised *
91	Original	125	First Revised *
92	Original	126	First Revised *
93	Original	127	First Revised *
94	Original	128	First Revised *
95	Original	129	First Revised *
96	Original	130	First Revised *
97	Original	131	Original
98	Original	132	Original

 Issued: April 10, 2007
 Effective: May 9, 2007

 In accordance with Case No. 00-21-TP-ACE, issued \_\_\_\_\_\_\_\_.
 June 20, 2002\_\_\_\_\_\_\_\_\_.

 Issued by:
 Louis F. McAlister, Jr., President

 Navigator Telecommunications, LLC.
 8525 Riverwood Park Drive, P.O. Box 13860

North Little Rock, AR 72113-0860

••

## CHECK SHEET

The Title Page and pages listed below of this tariff are effective as of the date shown. Revised sheets contain all changes from the original tariff that are in effect as of the date indicated.

PAGE	<u>REVISION</u>	PAGE	<b>REVISION</b>
133	First Revised	158	First Revised *
134	First Revised *	159	Original
135	First Revised *	160	Original
136	Second Revised *	161	Original
137	Second Revised *	162	First Revised *
138	First Revised *	163	First Revised *
139	Original	164	Original
140	First Revised *	165	Original
141	First Revised *	166	Original
142	First Revised *	167	Original
142.1	First Revised *	168	Original
142.2	First Revised *	169	Original
142.3	First Revised *	170	Original
142.4	First Revised *	171	First Revised *
142.5	Original	172	Original
142.6	Original	173	Original
142.7	Original	174	Original
142.8	Original	175	Original
142.9	Original	176	Original
143	Original	177	Original
144	Original	178	Original
145	Original	179	Original
146	Original	180	Original
147	Original	181	Original
148	Original	182	Original
149	Original	183	Original
150	Original	184	Original
151	Original	185	Original
152	Original	186	Original
153	Original	187	Original
154	Original	188	Original
155	First Revised	189	First Revised *
156	First Revised	190	Original
157	First Revised	191	Original

North Little Rock, AR 72113-0860

## CHECK SHEET

The Title Page and pages listed below of this tariff are effective as of the date shown. Revised sheets contain all changes from the original tariff that are in effect as of the date indicated.

<u>PAGE</u>	<u>REVISION</u>	PAGE	<b>REVISION</b>
192	Original		
193	Original		
194	First Revised *		
195	First Revised *		
195.1	First Revised *		
196	First Revised *		
197	First Revised *		
198	First Revised *		
199	First Revised *		
200	First Revised *		
201	Second Revised *		
202	Second Revised *		
203	First Revised *		
204	First Revised *		
205	First Revised *		
206	Second Revised *		
207	First Revised *		
208	First Revised *		
209	Second Revised *		
210	Second Revised *		
211	First Revised *		
212	First Revised *		
213	First Revised *		
214	First Revised		
215	First Revised		
216	First Revised		
217	First Revised *		
218	First Revised *		
219	First Revised *		
220	Second Revised *		
221	Original		
222	Original		
223	Original		

 Issued: April 10, 2007
 Effective: May 9, 2007

 In accordance with Case No. 00-21-TP-ACE, issued \_\_\_\_\_\_\_.
 June 20, 2002\_\_\_\_\_\_.

 Issued by:
 Louis F. McAlister, Jr., President

 Navigator Telecommunications, LLC.
 8525 Riverwood Park Drive, P.O. Box 13860

 North Little Rock, AR 72113-0860

## SECTION 3 - RATES – LOCAL EXCHANGE SERVICES, CONT.

## 3.2 Service Charges, cont. (1)(2)

#### 3.2.2 Nonrecurring Charges

A. Application of Nonrecurring Charges to Establish Services

Except where otherwise specified in this paragraph, the same nonrecurring charges apply to residence and non-residence service.

The Central Office Connection and Line Connection Charges do not apply to the transfer of an existing central office line servicing a reseller's end user when there is no interruption of service.

#### 1. Exchange Service

The following charges are applicable to each individual exchange service being established:

_	<u>Complex</u>	Simple
	<u>Maximum</u>	<u>Maximum</u>
a. Local Exchange Service		
- Service Ordering Charge		
per location, per occasion	n	
- Residence	\$25.50	\$25,50
- Non Residence	22.50	35.50
- Central Office Connection	on	
Charge, per termination		
- Residence	12.50	12.50
- Non Residence	25.00	18.50
- Line Connection Charge		
per termination		
- Residence	12.50	12.50
- Non Residence	22.50	32.50

(1) Current prices for these services are shown in Section 12.2 of this tariff.

(2) All minimum prices have been removed from this page.

Issued:	April 10, 2007	Effective: May 9, 2007
In accor	dance with Case No. 00-21-TP-ACE, issued	June 20, 2002 .
Issued b	y: Louis F. McAlister, Jr.,	President
	Navigator Telecommuni	cations, LLC.
	8525 Riverwood Park Drive,	P.O. Box 13860
	North Little Rock, AR 7	72113-0860

**(T)** 

(N)

# SECTION 3 - RATES – LOCAL EXCHANGE SERVICES, CONT.

#### **3.2** Service Charges, cont. (1)(2)

#### 3.2.2 Nonrecurring Charges

- A. Application of Nonrecurring Charges to Establish Service, cont.
  - 1. Exchange Service, cont.

The following charges are applicable to each individual exchange service being established:

	Complex	Simple
	Maximum	Maximum
b Central Office Connection		
Charge, per termination		
- Residence	-	-
- Non Residence	-	18.00
- Line Connection Charge,		
per termination (includes		
installation of telephone)		
- Residence	-	-
- Non Residence	-	74.50
c. Centrex stations and terminals		
Exhibition Hall stations, Airpo	ort	
Switching system stations		
- Service Ordering Charge		
per location, per occasion		
- Residence	-	-
- Non Residence	54.80	-
- Central Office Connection		
Charge, per station terminatio	n	
- Residence	-	-
- Non Residence	4.50	-
(1) Current prices for these services are shown in	Section 12.2 of this	tariff.

(2) All minimum prices have been removed from this page.

Issued: April 10, 2007 In accordance with Case No. 00-21-TP-ACE, issued <u>June 20, 2002</u>. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 **(T)** 

(N)

## SECTION 3 - RATES – LOCAL EXCHANGE SERVICES, CONT.

#### 3.2 Service Charges, cont. (1)(2)

## 3.2.2 Nonrecurring Charges, cont.

- A. Application of Nonrecurring Charges to Establish Service, cont.
  - 1. Exchange Service, cont.

Charges applicable to each individual exchange service being established:

	Complex	Simple	
	Maximum	Maximum	
- Line Connection Charge			
per station termination			
- Residence	-	-	
- Non Residence	5.50		
d. PBX Trunks			
- Service Ordering Charge			
per location, per occasion			
- Residence	32.50	-	
- Non Residence	32.50	-	
- Central Office Connection			
Charge, per termination			
- Residence	20.50	-	
- Non Residence	20.50	-	
- Line Connection Charge, per termination	13.35	-	

(1) Current prices for these services are shown in Section 12.2 of this tariff.

(2) All minimum prices have been removed from this page.

(N)

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 **(**T**)** 

# SECTION 3 - RATES - LOCAL EXCHANGE SERVICES, CONT.

#### 3.2 Service Charges, cont. (1)(2)

#### 3.2.2 Nonrecurring Charges

#### B. Application of Nonrecurring Charges for Restoral of Service

	Complex	Simple	
	Maximum	Maximum	
<ol> <li>Service restored after temporary denial but prior to completion of order to discontinue service, per service or system Restoral Charge</li> <li>Residence</li> </ol>	\$35.50	\$38.30	
- Non Residence	35.50	38.30	
<ol> <li>Service restored after temporary suspension Restoral Charge</li> <li>Residence</li> <li>Non Residence</li> </ol>	69.50 83.50	39.50 39.50	
<ul> <li>3. Service restored after temporary interception Restoral Charge</li> <li>Residence</li> </ul>	69.50 08.50	39.50 20.50	
<ul> <li>Non Residence</li> </ul>	98.50	39.50	

(1) Current prices for these services are shown in Section 12.2 of this tariff.

(2) All minimum prices have been removed from this page.

÷

(T)

(N)

### 3.3 Local Service Rates (1)(2)

#### 3.3.1 Flat Rate Service

Flat Rate service consists of a usage package that includes an unlimited number of local messages.

A.	Rates and Charges	Monthly Rate
	-	<u>Maximum</u>
	Residence PBX Trunks	\$11.50
	Residence Individual	\$10.00

#### 3.3.2 Local Service Rates

Description	Access Area	Network Access Line	Central Office Termination	Total Monthly Charge
NT TO 11		Maximum	Maximum	Maximum
Non - Residence				
- Non-Rotary	А	-	-	-
With Touch-Tone	B	17.95	4.30	22.25
	С	19.95	4.30	24.25
	D	22.45	4.30	26.75
- Rotary	А	-	-	_
- PBX Trunk	В	17.95	8.00	25.95
With Touch-Tone	- Ĉ	19.95	8.00	27.95
	D	22.45	8.00	30.45

A. Schedule of Monthly Rates – Network Access Lines

(1) Current prices for these services are shown in Section 12.3 of this tariff.

(2) All minimum prices have been removed from this page.

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 **(**T**)** 

••

#### 3.3 Local Service Rates, cont. (1)(2)

### 3.3.2 Local Service Rates, cont.

## A. Schedule of Monthly Rates – Network Access Lines, cont.

Description	Access Area	Network Access Line	Central Office Termination	Total Monthly <u>Charge</u>
		Maximum	Maximum	Maximum
Non - Residence				
- Non-Rotary	А	-	-	-
Without	В	17.95	4.30	22.25
Touch-Tone	С	19.70	4.30	24.00
	D	19.70	4.30	24.00
- Rotary	А	-	-	-
- PBX Trunk	В	17.95	8.00	25.95
Without	С	19.70	8.00	27.95
Touch-Tone	D	19.70	8.00	27.95
Residence				
- Individual	А	-	-	_
With Touch-Tone	В	6.40	4.30	10.70
	С	6.40	4.30	10.70
	D	6.40	4.30	10.70
- PBX Trunk	А	-	-	-
With Touch-Tone	В	17.95	8.00	25.95
	С	19.95	8.00	27.95
	D	22.45	8.00	30.45

(1) Current prices for these services are shown in Section 12.3 of this tariff.

(2) All minimum prices have been removed from this page.

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 (N)

••

# SECTION 3 - RATES – LOCAL EXCHANGE SERVICES, CONT.

# 3.3 Local Service Rates, cont. (1)(2)

# 3.3.2 Local Service Rates, cont.

# A. Schedule of Monthly Rates – Network Access Lines, cont.

Description	Access Area	Network Access Line	Central Office Termination	Total Monthly Charge
-		Maximum	Maximum	Maximum
Residence				
- Individual	А	-	-	-
Without	В	6.40	4.30	10.70
Touch-Tone	С	6.40	4.30	10.70
	D	6.40	4.30	10.70
- PBX Trunk	А	-	-	-
Without	В	17.95	8.00	25.95
Touch-Tone	С	19.70	8.00	27.70
	D	19.70	8.00	27.70

#### 3.3.3 Measured Rate Services

# A. Local Message Charge Schedule

Rate Mileage	Initial Minute or Fraction Thereof	Additional Minute or Fraction Thereof	
	Maximum.	<u> </u>	
0 - 10	\$.0453	\$.01	
11 - 22	.0497	.02	
23 and over	.0542	.02	

(1) Current prices for these services are shown in Section 12.3 of this tariff.

(2) All minimum prices have been removed from this page.

Issued:	April 10, 2007	Effective: May 9, 2007
In acco	rdance with Case No. 00-21-TP-ACE, issued	<u>June 20, 2002</u> .
Issued	by: Louis F. McAlister,	Jr., President
	Navigator Telecomm	unications, LLC.
	8525 Riverwood Park Driv	ve, P.O. Box 13860
	North Little Rock, A	R 72113-0860

**(T)** 

#### 3.3 Local Service Rates, cont. (1)(2)

#### 3.3.4 Local Calling Plus

Measured Rate Service shown above is not required to take advantage of Local Calling Plus. It is available to all measured, message and flat rate service residence and non-residence customers. Local usage rates are shown below.

#### A. Local Calling Plus Charge Schedule

	Initial Minute	Additional Minute
Rate Mileage	or Fraction Thereof	or Fraction Thereof
<u> </u>	Maximum	Maximum
0 - 10	\$.0490	\$.02
11 - 22	.0530	.025
23 and over	.0590	.03

(1) Current prices for these services are shown in Section 12.3 of this tariff.

(2) All minimum prices have been removed from this page.

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 **(**T**)** 

#### 3.4 Operator Service Rates – Traditional (1)(2)

3.4.1 Service Charges

	C C	Maximum Per Use
		Charge
	Automated Calling Card	\$1.00
	Station-to-Station	
	Customer Dialed-Operator	3.00
	Assisted-Calling Card	
	Station-to-Station	
	Operator Assisted	2.75
	Station-to-Station	
	Operator Assisted	6.00
	Person-to-Person	
	Operator Assisted	3.00
	Third Number Billed	
3.4.2	Directory Assistance Service - Local	
	Residence Charge	
	Where customer direct dials	\$.60
	directory assistance number	
	Where customer places call to directory	.90
	assistance attendant via an operator	
	Non - Residence Charge	
	Where customer direct dials	.60
	directory assistance number	
	Where customer places call to directory	.90
	assistance attendant via an operator	.90
	-	
	National Directory Assistance	1.50
	Direct Dialed	1.50
	Alternate Billed Calls	1.50

(1) Current prices for these services are shown in Section 12.4 of this tariff.

(2) All minimum prices have been removed from this page.

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 **(**T**)** 

#### **Operator Service Rates - Traditional, cont. (1)** 3.4

- 3.4.3 Information Call Completion (ICC) Service
  - A. Rates and charges are for automatic or operator call completion of a directory (T) assistance call after receipt of the requested number. If a customer selects an alternate billing option, i.e., calling card, billed-to-third party, collect and person-to-person, special handling, service charges as set forth above apply in addition to the ICC charge.

3.4.4 Busy Line Verify and Line Interrupt Service

(1) Current prices for these services are shown in Section 12.4 of this tariff.

(D)

(D)

# 3.5 Alternative Operator Service Rates (1)(2)

#### 3.5.1 Prices

## 1. Operator Assisted Services

	Maximum Nonrecurring
Description	Charge
Automated Calling Card Station-to-Station, Per call	\$ 0.70
Customer-Dialed Calling Card Station-to-Station, per call	1.70
Operator-Assisted Station-to-Station, Per call	2.50
Operator-Assisted Person-to-Person, Per call	4.80
Operator-Assisted Third Number Billed, Per call	2.50

(1) Current prices for these services are shown in Section 12.5 of this tariff.

(2) All minimum prices have been removed from this page.

(N)

Issued: April 10, 2007		Effective: May 9, 2007
In accordance with Case N	No. 00-21-TP-ACE, issued _	June 20, 2002
Issued by:	Louis F. McAlister, Jr.	, President
	Navigator Telecommun	ications, LLC.
	8525 Riverwood Park Drive	, P.O. Box 13860
	North Little Rock, AR	72113-0860

3.7	Call F	Blockir	ng	(T)
	3.7.1	Rates	s and Charges	
		A.	Current rates are as shown in Section 12.7.	(T)

(D)

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

### 3.8 Toll Restriction

**(T)** 

#### 3.8.1. Rates and Charges

Rates and charges apply to toll restriction service and are in addition to all other (T) rates and charges applicable to residence, non-residence, or PBX trunk service. See Section 3.3 for local service rates for these services.

A. Current rates are as shown in Section 12.8.

(T) (D)

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

### 3.9 Custom Calling Features (1)(2)

#### 3.9.1 Residence - Rates and Charges

	Maximum
	Per Month
Description /Billing Code/	Charge
Call Waiting /ESX/	\$9.40

#### 3.9.2 Business - Rates and Charges

	Maximum
	Per Month
Description /Billing Code/	Charge
Call Waiting /ESX/	\$18.20

(1) Current prices for these services are shown in Section 12.9 of this tariff.

(2) All minimum prices have been removed from this page.

(N)

**(T)** 

 Issued: April 10, 2007
 Effective: May 9, 2007

 In accordance with Case No. 00-21-TP-ACE, issued \_\_\_\_\_\_\_\_\_.
 June 20, 2002\_\_\_\_\_\_\_\_.

 Issued by:
 Louis F. McAlister, Jr., President

 Navigator Telecommunications, LLC.
 8525 Riverwood Park Drive, P.O. Box 13860

 North Little Rock, AR 72113-0860

**(T)** 

# SECTION 3 - RATES - LOCAL EXCHANGE SERVICES, CONT.

~

#### Advanced Custom Calling Features (1)(2) 3.10

3.10.1 Residence - Rates and Charges

Description /Billing Code/		Maximum Per Month <u>Charge</u>	
Caller ID /NSD/		9.00	(
			(
		Maximum	
Description		Per Use	
Description		Charge	(
Call Trace, per successful activation		4.75	
3.10.2 Non-Residence - Rates and Charges			
		Maximum	
Description /Billing Code/		Per Month Charge	
Caller ID /NSD/		9.00	
Call Waiting ID /NWT/	1.00	4.00	
		Maximum Per Use	
Description		Charge	
Call Trace, per successful activation		4.75	
te: The monthly rate for Caller ID with Name is in a	ddition to the mo	onthly rate for Caller ID.	
Current prices for these services are shown in S All minimum prices have been removed from the	ection 12.10 of		(
: April 10, 2007 ordance with Case No. 00-21-TP-ACE issued	Eff	ective: May 9, 2007	

Issued: A In accordance with Case No. 00-21-TP-ACE, issued \_\_\_\_\_June 20, 2002 Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

## 3.11 Central Office Services

3.11.1 Direct Inward Dialed (DID) Service

A. Regulations

DID service will be provided, subject to availability of central office facilities and telephone numbers in the serving central office. Customers utilizing DID must arrange to provide for interception of all unused telephone numbers in the groups of DID telephone numbers assigned by the Telephone Company.

B. Rates and Charges

Current prices for these services are shown in Section 12.11 of this tariff.

(T) (D)

#### 3.11 Central Office Services, cont.

3.11.2 Two-Way Direct Inward Dialing (DID) Service With Call Transfer, cont.

B. Regulations, cont.

- 7. A change in CO equipment could require the customer to discontinue the service or obtain service from another CO. The Company makes no guarantees and assumes no liability for loss of service to the customer, resulting from the conversion or upgrade of CO equipment.
- 8. Two-Way DID With Call Transfer works in conjunction with DID Service and the rates and charges specified in Paragraph C following are in addition to the rates and charges for DID Service specified in Paragraph 1.B of this section. Local and Message Toll charges are applicable.
- C. Rates and Charges
  - 1. Current rates are as shown in Section 12.11.2.

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 (T)

(T) (D)

#### 3.11 Central Office Services, cont.

3.11.2 Two-Way Direct Inward Dialing (DID) Service With Call Transfer, cont.

C. Rates and Charges, cont.

Current rates are as shown in Section 12.11.

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 (D)

(T) (D)

#### 3.12 Calling Party Number (CPN) Blocking (1),(2),(3),(4)

3.12.1 Per Line Calling Party Number Blocking (Calling Number Delivery Suppression)

The following rates and charges are applicable to Per Line Calling Party Number Blocking. See Section 2.6.2 for additional information related to this service.

Description	Nonrecurring Maximum Charge
Per Exchange Service, Residence/Non-Residence	\$12.30
Description	Monthly Maximum Charge
Per Exchange Service - Residence - Non-Residence	\$1.00 1.50

(1) Current prices for these services are shown in Section 12.12 of this tariff.

(2) The monthly rate for Blocking will not exceed the monthly rate for private listing service.

(3) Per Line Calling Party Number Blocking will be provided at no charge to qualified social service agencies, law enforcement organizations, and their certified employees, and volunteers. Per Line Calling Party Number Blocking will also be provided at no charge to customer-owned coin-operated telephone (COCOT) customers.

(4) All minimum prices have been removed from this page.

(N)

#### 3.13 SmartATM

#### 3.13.1 Description

- A. The SmartATM Service is only available to business subscribers and provides the following services (includes Touch-Tone):
  - 1. Business individual lines for Automated Teller Machines only.
  - 2. Unlimited calling to 800 toll-free numbers.
- B. The rate specified herein is subject to a 12-month service term.
- 3.13.2 Regulations and Service Limitations
  - A. Calls made outside the service description shown in Section 3.13.1 are subject to normal usage charges as described in other sections of this tariff.
  - B. Charges for this service commence when the company's information records are posted and are payable monthly in advance.
  - C. Charges for service are automatically discontinued upon service termination.
  - D. If service is terminated before the 12-month term, a charge of \$10 per (T) month will be applied for each line currently or previously classified as a SmartATM line. This charge will be imposed based upon the remainder of the commitment period.
  - E. Existing SmartATM subscribers may not take advantage of special promotions or rate changes prior to the end of the 12-month service term. This is applicable to existing lines only. New lines may qualify for any special promotions or price changes.

#### 3.13 SmartATM, cont.

A.

3.13.3	Rates and Charges		(T)

Current rates are as shown in Section 12.15. (T)

(D)

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

#### 3.14 Business - Bundled Service Offerings

- 3.14.1 SmartTime Unlimited for Business
  - A. Description
    - 1. This offering is available only to business customers (coin lines are not eligible) and provides for unlimited local and long distance calling. The customer may choose an unlimited number of features as shown in Section 3.9 of this tariff. Touch tone calling is included as part of this offering.
  - B. Regulations
    - 1. The customer must commit to at least a one-year term agreement to be eligible for the service.
    - 2. Maximum of ten (10) lines per location. Charges below are shown on a per line basis.
    - 3. Includes unlimited long distance calls for business customers, nondata, voice use only, within the 48 contiguous United States. Calls made to points outside of the contiguous 48 states, including all international calls, are not included in this offer, and will be billed at applicable rates. Subsequent to customer notification, Navigator may cancel subscriptions to any customer at any time due to fraud or misuse. Usage can be screened for other than voice usage through a review of call records for those customers identified as having significant usage. Taxes, fees, and any EAS charges also apply.
  - C. Rates and Charges
    - 1. Current rates are as shown in Section 12.16.1.

**(**T**)** 

**(D)** 

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

#### 3.14 Business - Bundled Service Offerings, cont.

- 3.14.2 SmartTime for Business
  - A. Description
    - This offering is available only to business customers (coin lines are not eligible) and provides for unlimited local calling. The customer may choose an unlimited number of features as shown in Section 3.9 of this tariff. Touch tone calling is included as part of this offering.

B. Regulations

- 1. The customer must commit to at least a one-year term agreement to be eligible for the service.
- 2. Long distance calls for business customers at \$.055 per minute is only for calls within the 48 contiguous United States. Calls made to points outside of the contiguous 48 states, including all international calls, are not included in this offer, and will be billed at applicable rates. Taxes, fees, and any EAS charges also apply.
- C. Rates and Charges
  - 1. Current rates are as shown in Section 12.16.2. (T)

**(D)** 

## SECTION 4 – PUBLIC TELEPHONE SERVICE, CONT.

#### 4.1 Independent Payphone Provider Services, cont.

- 1. Customer-Owned, Coin Operated Telephones (COCOT), cont.
  - E. Prices, cont.
    - 1. Service Elements, cont.
      - d. Answer Supervision Line Side

Current prices for these services are shown in Section 12.13 of this (T) tariff.

(D)

2. Other Applicable Charges and Payments

In addition to Telephone Exchange Service Network Access Rates, including Network Access Line Quantity Discount Rates, end user access line charges shown in Navigator's Telecommunications, LLC.'s Tariff F.C.C. No. 3 are also applicable.

Issued: April 10, 2007 In accordance with Case No. 00-21-TP-ACE, issued <u>June 20, 2002</u>. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 (D)

# SECTION 4 – PUBLIC TELEPHONE SERVICE, CONT.

#### 4.1 Independent Payphone Provider Services, cont.

2. COCOT - Coin Line, cont.

## D. Prices

1. Current prices for these services are shown in Section 12.13 of this tariff. **(T)** 

(D)

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued \_\_\_\_\_ June 20, 2002 Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

### SECTION 4 – PUBLIC TELEPHONE SERVICE, CONT.

#### 4.1 Independent Payphone Provider Services (1), cont.

2. COCOT - Coin Line, cont.

D. Prices, cont.

3. Other Optional Offerings

#### **Outgoing Only Service**

An optional service offered to customers who wish to maximize payphone usage or reduce the ability of end users to conduct business from a location that prohibits incoming calls. Callers hear a recording stating that the phone is not in service for incoming calls. It is the IPP's responsibility to state this restriction on the instruction card of the phone.

#### **Restricted Coin Access (RCA)**

An optional service for public telephones where a coin collecting device can be disabled during certain hours at the IPP's discretion to permit origination of Local calls, extended Local calls and Message Toll calls to only be charged to a third number, a Company calling card, or placed on a collect basis. Calls to public emergency numbers such as 911 will continue to be permitted and provided free of charge. End users placing coin calls during restricted hours will have their coins returned and will hear a recording that the phone is not equipped for coin calls during certain hours. It is the IPP's responsibility to state this restriction on the instruction card of the phone.

(D)

(1) Current prices for these services are shown in Section 12.13 of this tariff.

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued <u>June 20, 2002</u>. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

## SECTION 6 - RULES AND REGULATIONS – INTEREXCHANGE SERVICES, CONT.

#### 6.7 Billing and Payment for Service, cont.

6.7.3 If a Customer is not satisfied with the Company's response to an inquiry or request for credit, he or she may appeal to the Public Utilities Commission of Ohio for final resolution. The Public Utilities Commission of Ohio may be contacted at:

Public Utilities Commission of Ohio 180 E. Broad Street Columbus, OH 43215-3793 Toll-free telephone: 1-800-686-PUCO (7826)

#### 6.7.4 Delinquent Bills

Navigator Telecommunications, LLC. allows residential Customers at least 21 days from the date of the postmark of the bill to pay bill charges.

6.7.5 Late Payment Fees

The Company reserves the right to assess a one-time late payment fee of 1.5% per month on any past due balance. The late payment charge will not be applied to previous late payment charges that have been assessed but not yet been paid for, but will apply to the accumulated services for which the customer is in arrears. Late payment charges will be applied without discrimination.

6.7.6 Returned Check Charge

The Company will charge \$20 for each check returned for insufficient funds. This (D) charge may be waived under appropriate circumstances.

# SECTION 8 - MISCELLANEOUS SERVICES AND RATES – INTEREXCHANGE SERVICES

# 8.1 Directory Assistance (1)(2)

Directory Assistance is available to Customers of Navigator Telecommunications, LLC. Directory Assistance charges apply to each call to the Directory Assistance Bureau. Up to two requests may be made on each call to Directory Assistance. The Directory Assistance charge applies to each call regardless of whether the Directory Assistance Bureau is able to furnish the requested telephone number.

	Maximum
	Per Use
	Charge
Directory Assistance	\$1.75

(1) Current prices for these services are shown in Section 12.14 of this tariff.

(2) All minimum prices have been removed from this page.

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 **(**T**)** 

(T)

# **SECTION 12 – PRICE LIST**

#### 12.1 General

Rates shown in this section of the tariff are the actual prices currently being charged by Navigator Telecommunications, LLC. for the services shown. Minimum and maximum rates for these services are referenced in this section and are shown in other sections of this tariff.

### 12.2 Service Charges (1)

12.2.2 Nonrecurring Charges

A. Application of Nonrecurring Charges to Establish Services

Except where otherwise specified in this paragraph, the same nonrecurring charges apply to residence and non-residence service.

The Central Office Connection and Line Connection Charges do not apply to the transfer of an existing central office line servicing a reseller's end user when there is no interruption of service.

1. Exchange Service

The following charges are current rates applicable to each individual exchange service being established:

5 5	Com	plex	Sin	ple
		Non-		Non-
	Residence	Residence	<u>Residence</u>	<u>Residence</u>
a. Local Exchange Service	;			
- Scrvice Ordering				
Charge, per location, per occasion	\$17.65	\$15.85	\$17.65	\$25.50
- Central Office				
Connection Charge, per termination - Line connection charge	8.25	17.00	8.25	13.00
per termination	10.60	16.50	10.60	24.35

(1) Where required, maximum ranges for these services are shown in Section 3.2 of this tariff.

Issued: April 10, 2007 In accordance with Case No. 00-21-TP-ACE, issued <u>June 20, 2002</u>. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

...

# SECTION 12 – PRICE LIST, CONT.

# 12.2 Service Charges, cont. (1)

12.2.2 Nonrecurring Charges

- A. Application of Nonrecurring Charges to Establish Service, cont.
  - 1. Exchange Service, cont.

The following charges are current rates applicable to each individual exchange service being established:

č č	Com	plex	Sin	nple
		Non-		Non-
J	<u>Residence</u>	Residence	<u>Residence</u>	Residence
b Central Office				
Connection Charge,				
per termination	-	-	-	13.00
- Line Connection Charge				
per termination (include				
installation of telephone	e) -	-	-	63.00
c. Centrex stations and term	ninals			
Exhibition Hall stations,	Airport			
Switching system station	15			
- Service Ordering Charg	je			
per location, per occasion	on -	46.80	-	-
- Central Office Connecti	ion			
Charge, per station				
termination	-	1.75	-	-
- Line Connection Charge	e	0.05		
per station termination	-	3.35	-	-
d. PBX Trunks				
- Service Ordering Charg	;e			
per location, per occasio	on 24.50	24.50	-	-
- Central Office Connect	ion			
Charge, per termination		14.50	-	-
- Line Connection Charg				
per termination	10.35	10.35	-	-

(1) Where required, maximum ranges for these services are shown in Section 3.2 of this tariff.

Issued: April 10, 2007	Effective: May 9, 2007
In accordance with Case No.	00-21-TP-ACE, issued <u>June 20, 2002</u> .
Issued by:	Louis F. McAlister, Jr., President
	Navigator Telecommunications, LLC.
852	5 Riverwood Park Drive, P.O. Box 13860
	North Little Rock, AR 72113-0860

#### 12.2 Service Charges, cont. (1)

12.2.2 Nonrecurring Charges

- A. Application of Nonrecurring Charges to Establish Service, cont.
  - 1. Exchange Service, cont.

The following charges are current rates applicable to each individual exchange service being established:

	Com	plex	Sin	<u>nple</u>
		Non-		Non-
	Residence	<b>Residence</b>	<u>Residence</u>	<u>Residence</u>
e.	Misc. Services/Features (2)			
	- Service Ordering Charge			
	per location, per occasion 7.95	7.30	6.00	7.30

- (1) Where required, maximum ranges for these services are shown in Section 3.2 of this tariff (T)
- (2) Miscellanous Services and Features charges apply per individual exchange service, per occasion to Custom Calling Service and Advanced Custom Calling Service (other than Calling Party Number Blocking or Call Trace) excluding Call Waiting and Caller ID.

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 . Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

#### 12.2 Service Charges, cont. (1)

12.2.2 Nonrecurring Charges

A. Application of Nonrecurring Charges to Establish Service, cont.

2. Network Wiring Charge

		Nonrecurring
		<u>Charge</u>
-	Per semi-public telephone	\$86.50
-	Change to/from handset with	75.55
	volume control	

#### B. Application of Nonrecurring Charges for Restoral of Service

	Complex		Sim	
	Residence	Non- <u>Residence</u>	Residence	Non- <u>Residence</u>
<ol> <li>Service restored after temporary denial but prior to completion of order to discontinue service, per service or system</li> </ol>				
Restoral Charge	\$26.55	\$26.55	\$32.30	\$32.30
2. Service restored after temporary suspension				
Restoral Charge	62.30	75.85	33.55	33.55
3. Service restored after temporary interception				
Restoral Charge	62.50	87.70	33.55	33.55

(1) Where required, maximum ranges for these services are shown in Section 3.2 of this tariff. (T)

#### Local Service Rates (1) 12.3

## 12.3.1 Flat Rate Service

Flat Rate service consists of a usage package that includes an unlimited number of local messages.

Rates and Charges Α.

	Usage Package Monthly Rate
Residence PBX Trunks	\$ 8.95
<b>Residence Individual</b>	\$ 7.55

# 12.3.2 Local Service Rates

#### Schedule of Monthly Rates - Network Access Lines A.

Description	Access Area	Network Access Line	Central Office Termination	Total Monthly <u>Rate</u>	
Non - Residence					
- Non-Rotary	А	Future Use	-	-	
With Touch-Tone	В	\$15.95	\$2.30	\$18.25	
	С	17.95	2.30	20.25	
	D	20.45	2.30	22.75	
- Rotary or	А	Future Use	-	-	
- PBX Trunk	В	\$15.95	\$6.00	\$21.95	
With Touch-Tone	С	17.95	6.00	23.95	
	D	20.45	6.00	26.45	

(T) (1) Where required, maximum ranges for these services are shown in Section 3.3 of this tariff.

Issued: April 10, 2007	Effective: May 9, 2007
In accordance with Case N	o. 00-21-TP-ACE, issued <u>June 20, 2002</u> .
Issued by:	Louis F. McAlister, Jr., President
	Navigator Telecommunications, LLC.
8	525 Riverwood Park Drive, P.O. Box 13860
	North Little Rock, AR 72113-0860

# 12.3 Local Service Rates, cont. (1)

12.3.2 Local Service Rates, cont.

# A. Schedule of Monthly Rates – Network Access Lines, cont.

Description	Access Area	Network Access Line	Central Office Termination	Total Monthly Rate
Non - Residence				
- Non-Rotary	А	Future Use	-	-
Without	В	\$15.95	\$2.30	\$18.25
Touch-Tone	С	17.70	2.30	20.00
	D	17.70	2.30	20.00
- Rotary or	А	Future Use	-	-
- PBX Trunk	В	\$15.95	\$6.00	\$21.95
Without	С	17.70	6.00	23.70
Touch-Tone	D	17.70	6.00	23.70
Residence				
- Individual	А	Future Use	-	-
With Touch-Tone	В	\$4.40	\$2.30	\$6.70
	С	4.40	2.30	6.70
	D	4.40	2.30	6.70
- PBX Trunk	A	Future Use	-	•
With Touch-Tone	В	\$15.95	\$6.00	\$21.95
	С	17.95	6.00	23.95
	D	20.45	6.00	26.45

(1) Where required, maximum ranges for these services are shown in Section 3.3 of this tariff.

Issued: April 10, 2007	Effective: May 9, 2007
In accordance with Case Ne	o. 00-21-TP-ACE, issued <u>June 20, 2002</u> .
Issued by:	Louis F. McAlister, Jr., President
	Navigator Telecommunications, LLC.
8	525 Riverwood Park Drive, P.O. Box 13860
	North Little Rock, AR 72113-0860

# 12.3 Local Service Rates, cont. (1)

12.3.2 Local Service Rates, cont.

# A. Schedule of Monthly Rates – Network Access Lines, cont.

Description	Access Area	Network Access Line	Central Office Termination	Total Monthly Rat <del>e</del>	
Residence					
- Individual	А	Future Use	~	-	
Without	В	\$4.40	\$2.30	\$6.70	
Touch-Tone	С	4.40	2.30	6.70	
	D	4.40	2.30	6.70	
- PBX Trunk	А	Future Use	-	-	
Without	В	\$15.95	\$6.00	\$21.95	
Touch-Tone	С	17.70	6.00	23.70	
	D	17.70	6.00	23.70	

# 12.3.3 Measured Rate Services

A. Local Message Charge Schedule

	Initial Minute	Additional Minute
Rate Mileage	or Fraction Thereof	or Fraction Thereof
0 - 10	\$.0353	\$.0088
11 - 22	.0397	.0132
23 and over	.0442	.0177

(1) Where required, maximum ranges for these services are shown in Section 3.3 of this tariff. (T)

Issued: April 10, 2007	Effective: May 9, 2007	
In accordance with Case No. 00-21-TP-ACE, issued	<u>June 20, 2002</u> .	
Issued by: Louis F. McAlister, Jr	., President	
Navigator Telecommun	ications, LLC.	
8525 Riverwood Park Drive, P.O. Box 13860		
North Little Rock, AR 72113-0860		

**(T)** 

## SECTION 12 – PRICE LIST, CONT.

#### 12.3 Local Service Rates, cont. (1)

12.3.4 Local Calling Plus

# A. Local Calling Plus Charge Schedule

	Initial Minute	Additional Minute
Rate Mileage	or Fraction Thereof	or Fraction Thereof
0 - 10	\$.0390	\$.01
11 - 22	.0430	.015
23 and over	.0490	.02

(1) Where required, maximum rates for these services are shown in Section 3.3 of this tariff.

Issued: April 10, 2007 In accordance with Case No. 00-21-TP-ACE, issued <u>June 20, 2002</u>. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

# 12.4 Operator Service Rates – Traditional (1)

12.4.1 Service Charges

12.1.1 Service Charges	Per Use	
	Charge	
Automated Calling Card	\$.95	
Station-to-Station		
Customer Dialed-Operator	1.95	
Assisted-Calling Card		
Station-to-Station		
Operator Assisted	1.75	
Station-to-Station		
Operator Assisted	3.75	
Person-to-Person		
Operator Assisted	2.25	
Third Number Billed		
12.4.2 Directory Assistance Service - Local		
Residence Charge		
Where customer direct dials	\$1.10	
directory assistance number		
Where customer requests operator assis call to directory assistance, the surcha	-	
Non - Residence Charge		
Where customer direct dials	1.10	
directory assistance number		
Where customer requests operator assis call to directory assistance, the surcha	•	
National Directory Assistance		
Direct Dialed	1.50	
Alternate Billed Calls	1.50	
(1) Where required, maximum rates for these services a	re shown in Section 3.4 of this tariff.	(T)
Issued: April 10, 2007	Effective: May 9, 2007	_

Issued: April 10, 2007	Effec	tive: May 9, 2007
In accordance with Case No.	00-21-TP-ACE, issuedJune 20, 2002	<u> </u> .
Issued by:	Louis F. McAlister, Jr., President	
	Navigator Telecommunications, LLC.	
85	25 Riverwood Park Drive, P.O. Box 13860	
	North Little Rock, AR 72113-0860	

# 12.4 Operator Service Rates – Traditional, cont. (1)

12.4.3 Information Call Completion (ICC) Service

	Per Use
	Charge
ICC, per call	.25

#### 12.4.4 Busy Line Verify and Line Interrupt Service

	Per Use
	Charge
Verification	1.75
Interrupt	2.00

 Issued:
 April 10, 2007
 Effective:
 May 9, 2007

 In accordance with Case No. 00-21-TP-ACE, issued
 June 20, 2002
 .

 Issued by:
 Louis F. McAlister, Jr., President

 Navigator Telecommunications, LLC.
 8525 Riverwood Park Drive, P.O. Box 13860

 North Little Rock, AR 72113-0860

(D)

# 12.5 Alternative Operator Service Rates (1)

# 12.5.1 Prices

# 1. Operator Assisted Services

Description	Nonrecurring Charge
Automated Calling Card Station-to-Station, Per call	\$0.70
Customer-Dialed Calling Card Station-to-Station, per call	1.70
Operator-Assisted Station-to-Station, Per call	2.50
Operator-Assisted Person-to-Person, Per call	4.80
Operator-Assisted Third Number Billed, Per call	2.50

(1) Where required, maximum rates for these services are shown in Section 3.5 of this tariff. (T)

Issued: April 10, 2007	Effective: May 9, 2007
In accordance with Case No. 00-21-TP-ACE, issued _	June 20, 2002
Issued by: Louis F. McAlister, Jr	., President
Navigator Telecommun	ications, LLC.
8525 Riverwood Park Drive	, P.O. Box 13860
North Little Rock, AR	72113-0860

#### 12.5 Alternative Operator Service Rates, cont.

#### 12.5.1 Prices

2. Message Toll Service Rates (1), per minute

••

	<u>Day</u> ]	Artes Evening Rates Night & W		Night & Wo	eekend Rates		
Rate	Initial	Add'l	Initial	Add'l	Initial	Add'l	
<u>Mileage</u>	Minute	Minute	Minute	Minute	Minute	Minute	_
<u>Schedule A (2)</u>							C (
1-10	\$.23	\$.23	\$.16	\$.16	\$.16	\$.16	
11-22	.23	.23	.16	.16	.16	.16	
22-55	.23	.23	.16	.16	.16	.16	
56-124	.23	.23	.16	.16	.16	.16	
125-End	.23	.23	.16	.16	.16	.16	
Schedule B (2)							ſ
1-10	\$.23	\$.23	\$.16	\$.16	\$.16	\$.16	
11-22	.23	.23	.16	.16	.16	.16	
22-55	.23	.23	.16	.16	.16	.16	
56-124	.23	.23	.16	.16	.16	.16	
125-End	.23	.23	.16	.16	.16	.13	
Operator Assisted C	Calls – Inmate	Services (3)					ſ
1-10	\$.25	\$.16	\$.12	\$.06	\$.12	\$.06	``
11-22	.28	.16	.160	.10	.160	.10	
22-55	.28	.20	.160	.10	.160	.10	
56-124	.28	.20	.160	.10	.160	.10	
125-End	.28	.20	.160	.10	.160	.10	
) Day Rates – Monday thru Friday 8:00 AM to but not including 5:00 PM							

- Day Rates Monday thru Friday 8:00 AM to but not including 5:00 PM Evening Rates – Monday thru Friday 5:00 PM to but not including 11:00 PM Night & Weckend Rates – Monday thru Friday 11:00 PM to but not including 8:00 AM Saturday, Sunday, and Holidays – All Day
- (2) Schedule A rates apply to calls terminating/originating in an Independent Telephone (T) Company Exchange. Schedule B rates apply for customer-dialed station-to-station calling.
- (3) Maximum rates are as shown in the PUCO's Service Requirements Form.

(T) (D)

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 **(**T**)** 

# 12.5 Alternative Operator Service Rates, cont.

12.5.1 Prices

2. Message Toll Service Rates, cont.

MTS Rate Discounts

Description	Discount
\$ 0.00 - \$25.00	0%
25.00 - 100.00	10%
100.01 - 200.00	15%
200.01 and up	20%

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 . Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 (D)

#### **12.6** Directory Services (1)

12.6.1 Directory Listings

The following monthly rates apply to both regular and special types of additional listings:

Non-residence, additional listing	Monthly <u>Charge</u> \$3.00	<u>USOC</u> CLT
Non-residence, foreign listing	3.00	FLT
Residence, additional listing	.90	RLT

#### 12.6.2 Private and Semi-Private Listing Service

Private listing service, each service	Monthly <u>Charge</u> 2.10	<u>USOC</u> NPU
Semi-private listing service, each service	2.10	NLT
Optional arrangement for relaying messages to private listing service customers, each message	1.80	-

(1) Where required, maximum rates for these services are shown in Section 3.6 of this tariff. (T)

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

#### 12.7 Call Blocking

**(**T**)** 

(D)

12.7.1 Rates and Charges

1. Residence and Non-Residence Customer Requested Call Blocking

The following rates and charges will be waived for the initial establishment of call blocking. These charges will apply to subsequent requests for blocking.

	Nonrecurring Charge	USOC
a. Residence Service		
Call blocking, per service	\$ 9.30	RTVXN
b. Non-Residence Service		
Call blocking, per individual		
or trunk or WATS service	\$ 9.30	RTVXN
c. Centrex Service		
Call blocking, Centrex stations	\$41.55	RTVXN
2. Sponsor Requested Call Blocking		
The following rates and charges apply	y to sponsor requested	l call blocking:
a. Rèsidence Service		
Call blocking, per request,		
per service	\$60.00	NR9CB
b. Non-Residence Service		
Call blocking per request, per		
individual or trunk		
or WATS service	\$60.00	NR9CB
c. Centrex Service		
Call blocking, per request, per		
Centrex system	\$60.00	NR9CB

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 . Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

ι...

#### 12.8 Toll Restriction

**(T)** 

(D)

## 12.8.1 Rates and Charges

The following rates and charges apply to toll restriction service and are in addition to all other rates and charges applicable to residence, non-residence, or PBX trunk service. See Section 3.4 for local service rates for these services.

Toll Restriction, per individual residence service	Nonrecurring <u>Charge</u> \$12.00	<u>USOC</u> RTVX5
Toll Restriction, per individual non-residence service or PBX trunk equipped	\$24.40	КХТ
Monthly Charges Toll Restriction, per individual residence service	Monthly <u>Rate</u> \$2.90	<u>USOC</u> RTVX5
Toll Restriction, per individual non-residence service or PBX trunk equipped	\$51.70	КХТ

Note: A toll restriction charge will not apply to those residential customers: 1) who elect toll restriction in lieu of a deposit, 2) who elect toll restriction while making payments for a final bill, or 3) who elect payment arrangements and toll restriction in lieu of denial.

Issued: April 10, 2007		Effective: May 9, 2007
In accordance with Case	No. 00-21-TP-ACE, issued _	June 20, 2002
Issued by:	Louis F. McAlister, Jr	., President
	Navigator Telecommun	ications, LLC.
	8525 Riverwood Park Drive	, P.O. Box 13860
	North Little Rock, AR	72113-0860

# 12.9 Custom Calling Features

12.9.1 Residence - Rates and Charges

Description /Billing Code/	Monthly Charge Per Line
Call Waiting /ESX/	\$4.15
Call Forwarding /ESM/	4.00
Call Forwarding /TAS/CFW/	4.00
Three-Way Calling /ESC/	4.00
Description /Billing Code/	Per Use Charge
Three-Way Calling	\$.70

# 12.9.2 Business - Rates and Charges

	Monthly Charge
Description /Billing Code/	Per Line
Call Waiting /ESX/	\$6.00
Call Forwarding /ESM/	7.00
Call Forwarding /TAS/CFW/	6.50
Three-Way Calling /ESC/	6.50
Speed Calling 8/ESL	6.00
Speed Calling 30/ESF	7.00
Description /Billing Code/	Per Use Charge
Three-Way Calling	\$1.99

.

(1) Where required, maximum rates for these services are shown in Section 3.9 of this tariff. (T)

Issued: April 10, 2007	Effective: May 9, 2007
In accordance with Case No. 00-21-TP-ACE, issued _	June 20, 2002 .
Issued by: Louis F. McAlister, Jr	., President
Navigator Telecommun	nications, LLC.
8525 Riverwood Park Drive	e, P.O. Box 13860
North Little Rock, AR	. 72113-0860

## 12.10 Advanced Custom Calling Features (1)

#### 12.10.1 Residence - Rates and Charges

	Monthly Charge
Description /Billing Code/	Per Line
Repeat Dialing /NSS/	\$4.10
Caller ID /NSD/	6.00
Caller ID w/Name /NMP/ (See Note)	1.95
Automatic Callback /NSQ/	4.10
Call Screening /NSY/	4.00
Description	Per Use Charge
Repeat Dialing	\$.75
Automatic Callback	.70
Call Trace, per successful activation	3.50
12.10.2 Non-Residence - Rates and Charges	
-	Monthly Charge
Description /Billing Code/	Per Line
Repeat Dialing /NSS/	\$6.50
Caller ID /NSD/	7.00
Caller ID w/Name /NMP/ (See Note)	4.20
Automatic Callback /NSQ/	6.00
Call Screening /NSY/	4.00
Call Waiting ID/NWT/	2.00
Description	Per Use Charge
Repeat Dialing	\$.75
Automatic Callback	1.99
Call Trace, per successful activation	4.75

Note: The monthly rate for Caller ID with Name is in addition to the monthly rate for Caller ID.

(1) Where required, maximum rates for these services are shown in Section 3.10 of this tariff. (T)

#### 12.11 Central Office Services

**(T)** 

### 12.11.1 Direct Inward Dialed (DID) Service

#### A. Rates and Charges

The following rates and charges are in addition to rates and charges applicable for PBX trunks:

Des	scription /Billing Code/	Nonrecurring Charge	Monthly Charge	
1.	Each group of 20 DID numbers or fraction thereof /ND4/ Or	\$150.00(1)	\$3.20	(T)
	Each individual DID number /SOT/	-	.25	
2.	Each DID trunk termination in central office /NDT/	155.00	19.00	

(D)

(1) In lieu of the nonrecurring charge included in Section 3.11.1.B.1, the nonrecurring charge (T) per additional group of 20 DID numbers provided on the same occasion as an initial group of 20 DID numbers is \$54.20.

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued <u>June 20, 2002</u>. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

#### 12.11 Central Office Services, cont.

## 12.11.2 Two-Way Direct Inward Dialing (DID) Service With Call Transfer

...

#### A. Rates and Charges

Service Establishment

Description/Billing Code/	Nonrecurring Charge	Monthly Charge
1. With initial DID Service, per Trunk Group /SEPD1/	\$175.00	-
2.Subsequent to Establishment of DID service, per Trunk Group /SEPD1/	175.00	-
3.Two-Way DID Trunk With Call Transfer / 3CW/	-	23.00
Conversions		
1.Of entire DID Trunk Group to Two-Way DID or entire Two-Way DID Trunk Group to DID /NR9DD/	300.00	-
2.Of individual DID Trunks to a new Two-Way DID Trunk Group, per Trunk Group /SEPD1/	175.00	-
3.Two-Way DID Trunk with Call Transfer /3CW/	-	23.00
4. Change in Outpulsing, Start Dial, or Signal Type, per Trunk Group /REAJA/	41.55	-
5. Change or Redesign in Signaling or Transmission Interface, per occurrence /REAJB/	150.00	-

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued \_\_\_\_\_\_June 20, 2002\_\_\_\_\_. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860

### 12.12 Calling Party Number (CPN) Blocking (1)

12.12.1 Per Line Calling Party Number Blocking (Calling Number Delivery Suppression)

The following rates and charges are applicable to Per Line Calling Party Number Blocking. See Section 2.6.2 for additional information related to this service.

Description	Nonrecurring Charge	Monthly <u>Charge</u>
Per Exchange Service - Residence	\$9.30	\$.50
- Non-Residence	9.30	1.00

(1) Where required, maximum rates for these services are shown in Section 3.12 of this tariff. (T)

#### 12.13 Independent Payphone Provider Services, cont.

12.13.1 Customer-Owned, Coin Operated Telephones (COCOT), cont.

A. Prices, cont.

1. Service Elements, cont.

d. Answer Supervision – Line Side

	Nonrecurring	Monthly	
Description	Charge	Charge	_
Answer Supervision equipped with line side			
Interface, per line	\$2.50	\$1.60	

2. Other Applicable Charges and Payments

In addition to Telephone Exchange Service Network Access Rates, including Network Access Line Quantity Discount Rates, end user access line charges shown in Navigator's Telecommunications, LLC.'s Tariff F.C.C. No. 3 are also applicable.

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 . Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 (D)

# 12.13 Independent Payphone Provider Services, cont. (1)

- 2. COCOT Coin Line
  - A. Prices
    - 1. Service Elements

Description	Access Area	Monthly Charge
Telephone Exchange Service Network Access Rates		
Network Access, per Coin Line	А	-
Network Access, per Coin Line	В	\$25.45
Network Access, per Coin Line	С	27.45
Network Access, per Coin Line	D	29.95

2. Other Applicable Charges and Payments

In addition to the above rates, interstate End User Common Line (EUCL) charges are also applicable.

Issued: April 10, 2007 In accordance with Case No. 00-21-TP-ACE, issued <u>June 20, 2002</u>. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 (D)

(D)

# 12.13 Independent Payphone Provider Services, cont.

- 2. COCOT Coin Line, cont.
  - A. Prices, cont.

3.	Other	Optional	Offerings
----	-------	----------	-----------

	Nonrecurring	Monthly	
Description	Charge	Charge	
Outgoing Only Service	-	-	
Restricted Coin Access (RCA)	\$212.00	\$10.75	

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002. Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860 (D)

### 12.14 Directory Assistance – Interexchange Services

Directory Assistance is available to Customers of Navigator Telecommunications, LLC. Directory Assistance charges apply to each call to the Directory Assistance Bureau. Up to two requests may be made on each call to Directory Assistance. The Directory Assistance charge applies to each call regardless of whether the Directory Assistance Bureau is able to furnish the requested telephone number.

	Per Use
	<u>Charge</u>
Directory Assistance	\$1.25

Issued: April 10, 2007 Effective: May 9, 2007 In accordance with Case No. 00-21-TP-ACE, issued June 20, 2002 . Issued by: Louis F. McAlister, Jr., President Navigator Telecommunications, LLC. 8525 Riverwood Park Drive, P.O. Box 13860 North Little Rock, AR 72113-0860